brought to you by TCORE



EUROPEAN CENTRAL BANK

WORKING PAPER SERIES NO. 498 / JUNE 2005

ECB-CFS RESEARCH NETWORK ON CAPITAL MARKETS AND FINANCIAL INTEGRATION IN EUROPE

FINANCIAL INTEGRATION AND ENTREPRENEURIAL ACTIVITY

EVIDENCE FROM
FOREIGN BANK ENTRY IN
EMERGING MARKETS

by Mariassunta Giannetti and Steven Ongena













In 2005 all ECB publications will feature a motif taken from the €50 banknote.

WORKING PAPER SERIES

NO. 498 / JUNE 2005

ECB-CFS RESEARCH NETWORK ON CAPITAL MARKETS AND FINANCIAL INTEGRATION IN EUROPE

FINANCIAL INTEGRATION AND ENTREPRENEURIAL ACTIVITY

EVIDENCE FROM FOREIGN BANK ENTRY IN EMERGING MARKETS '

by Mariassunta Giannetti ² and Steven Ongena ³

This paper can be downloaded without charge from http://www.ecb.int or from the Social Science Research Network electronic library at http://ssrn.com/abstract_id=726692.

Telephone: +31 13 4662417, Fax: +31 13 4662875, e-mail: steven.ongena@tilburguniversity.nl.



- I We are grateful to Gunseli Tumer Alkan, Ralph Koijen, Evren Ors, Fabiana Penas and seminar participants at CentER Tilburg University, HEC (Paris) and the Universities of Amsterdam, Frankfurt and Munich for comments. Marina Martinova and Luc Renneboog kindly shared data. Lingxiao Qu provided research assistance. Giannetti gratefully acknowledges financial support from the European Central Bank, under the Lamfalussy Fellowship Program, and the Bankforskningsinstitutet. The views of this paper are the authors' and do not reflect those of the ECB, the Eurosystem, or its staff.
 - 2 Stockholm School of Economics, ECGI, and CEPR, Department of Finance and SITE, PO Box 6501, S 11 383 Stockholm, Sweden; Telephone: +46 8 7369607, Fax: +46 8 316422, e-mail: mariassunta.giannetti@hhs.se.
 3 CentER – Tilburg University and CEPR, Department of Finance, PO Box 90153, NL 5000 LE Tilburg, The Netherlands;

ECB-CFS Research Network on "Capital Markets and Financial Integration in Europe"

This paper is part of the research conducted under the ECB-CFS Research Network on "Capital Markets and Financial Integration in Europe". The Network aims at stimulating top-level and policy-relevant research, significantly contributing to the understanding of the current and future structure and integration of the financial system in Europe and its international linkages with the United States and Japan. After two years of work, the ECB Working Paper Series is issuing a selection of papers from the Network. This selection is covering the priority areas "European bond markets", "European securities settlement systems", "Bank competition and the geographical scope of banking activities", "international portfolio choices and asset market linkages" and "start-up financing". It also covers papers addressing the impact of the euro on financing structures and the cost of capital.

The Network brings together researchers from academia and from policy institutions. It has been guided by a Steering Committee composed of Franklin Allen (University of Pennsylvania), Giancarlo Corsetti (European University Institute), Jean-Pierre Danthine (University of Lausanne), Vitor Gaspar (ECB), Philipp Hartmann (ECB), Jan Pieter Krahnen (Center for Financial Studies) and Axel Weber (CFS). Mario Roberto Billi, Bernd Kaltenhäuser (both CFS), Simone Manganelli and Cyril Monnet (both ECB) supported the Steering Committee in its work. Jutta Heeg (CFS) and Sabine Wiedemann (ECB) provided administrative assistance in collaboration with staff of National Central Banks acting as hosts of Network events. Further information about the Network can be found at http://www.eu-financial-system.org.

© European Central Bank, 2005

Address

Kaiserstrasse 29 60311 Frankfurt am Main, Germany

Postal address

Postfach 16 03 19 60066 Frankfurt am Main, Germany

Telephone

+49 69 1344 0

Internet

http://www.ecb.int

Fax

+49 69 1344 6000

411 144 ecb d

All rights reserved.

Reproduction for educational and noncommercial purposes is permitted provided that the source is acknowledged.

The views expressed in this paper do not necessarily reflect those of the European Central Bank.

The statement of purpose for the ECB Working Paper Series is available from the ECB website, http://www.ecb.int.

ISSN 1561-0810 (print) ISSN 1725-2806 (online)

CONTENTS

Abstract	4
I Introduction	5
II Theoretical predictions on the effects bank entry in Eastern European econor	_
III Methodology and identification	15
IV Data and sample characteristics	18
V Results	22
VI Conclusion	33
References	35
Tables	38
Notes	48
European Central Bank working paper ser	ries 49

Abstract

An extensive empirical literature has documented the positive growth effects of equity market liberalization. However, this line of research ignores the impact of financial integration on a category of firms crucial for economic development, i.e. the small entrepreneurial firms. This paper aims to fill this void. We employ a large panel containing almost 60,000 firm—year observations on listed and unlisted companies in Eastern European economies to assess the differential impact of foreign bank lending on firm growth and financing. Foreign lending stimulates growth in firm sales, assets, and leverage, but the effect is dampened for small firms. We also find that firms started during the transition period of 1989-1993 — arguably the most connected businesses — benefit least from foreign bank entry. This finding suggests that foreign banks can help mitigate connected lending problems and improve capital allocation.

Keywords: foreign bank lending, emerging markets, competition, lending relationships. **JEL**: G21, L11, L14.

I. Introduction

Neoclassical theory predicts that financial integration can foster growth in emerging markets because it permits capital from rich countries to be invested in economies with low savings but high growth opportunities. Empirical work has focused so far on the impact of equity market liberalization on growth. Henry (2000a, b, 2003) and Bekaert, Harvey and Lundblad (2003) among others show that equity market liberalization decreases the cost of capital, causes investment booms, and increases aggregate growth. Recent empirical firm-level evidence corroborates and extends these aggregate findings. Chari and Henry (2004) for example show that stock prices rally following equity market liberalization. They also document that companies with a larger free float and more liquid stocks tend to attract more investor interest and experience a larger decrease in their cost of equity than the other listed companies.

While listed companies seemingly benefit from financial integration through a lower cost of equity capital, the impact of integration on non-listed firms has not been investigated thoroughly yet and hence remains unclear. In developing countries stock markets are often not well developed and as a consequence few firms are listed (La Porta, Lopez-de-Silanes, Shleifer and Vishny (1998)). Growth prospects in those countries depend to a large extent on the creation of new businesses and investment of non-listed companies.

This paper aims to analyze how and to what extent the process of financial integration can benefit this category of small entrepreneurial firms, an issue that has so far been largely neglected in the literature. In order to do so, we focus on a different aspect of financial integration, which has captured a lot of attention in the policy debate, but less so in the academic community: foreign bank entry.

Unlisted companies in countries with underdeveloped equity markets and weak shareholder protection rely to a large extent on debt and specifically on bank credit to fund investment (Booth, Aivazian, Demirguc-Kunt and Maksimovic (2001) and Giannetti (2003)). Foreign banks may thus represent an invaluable source of capital for small firms and foster the creation of new companies.

Foreign banks may not only have easier access to foreign capital than domestic banks, and thus present a stable source of external funds for firms, but they may also contribute to mitigating problems that afflict bank lending. In many developing countries, banks often lend to cronies (Laeven (2001) and La Porta, Lopez-de-Silanes and Zamarripa (2003)). As a consequence established companies owned by related individuals receive funding even if inefficient, while young and potentially highly profitable firms face credit rationing. Foreign banks have fewer connections to local families and politicians. Therefore, foreign banks may be more inclined to fund promising projects, rather than related or state-owned firms. In addition, foreign banks may import lending expertise and sound practices.

There are reasons however why small firms may not be able to benefit to the full extent from financial integration, even in the case of foreign bank entry. Foreign banks may lack local information; a major problem in countries where asymmetric information problems are severe and legal enforcement is weak (Acharya, Sundaram and John (2004)). In addition foreign banks are often large organizations and reluctant to decentralize decision power. However decentralization is necessary if lending decisions need to be based on soft information, as is often the case when dealing with small and young firms. As a result the local branches of foreign banks may specialize in funding large firms and overlook small firms. Such neglect may create concerns that foreign bank presence may be detrimental to the financing and growth of small and young businesses, if foreign banks would compete away domestic banks. To conclude, small and young firms may be able to benefit from financial integration but even if financial integration involves foreign bank entry, possibly only to a lesser extent than large and established companies. To the best of our knowledge, so far no other study has investigated this differential impact of integration.

We explore a comprehensive dataset containing both listed and unlisted companies operating in the Eastern European economies. The dataset we employ is the most comprehensive source of information on entrepreneurial companies in emerging markets. The large panel, containing almost 60,000 firm—year observations, allows us to assess the differential impact of foreign bank lending on firm growth and financing. We face a potentially insidious endogeneity problem, i.e. foreign banks may in particular enter countries that are expected to grow more. We instrument our

proxies for foreign bank presence with characteristics of the institutional environment that are known to affect foreign banks' willingness to grant loans but are predetermined with respect to foreign bank entry. Additionally, we are not only studying the effect of foreign lending on average firm growth, but also investigate which type of firms grows more. This investigation significantly assuages any lingering doubts about the direction of causality.

In short, we find that foreign lending stimulates growth in firm sales, assets, and leverage, but that the effect is significantly dampened for small firms. Our findings suggest that although large firms benefit more from foreign bank presence, small entrepreneurial companies also profit from financial integration.

Since we focus on Eastern European economies, we can use the regime shift that took place between 1989-1993 as a natural experiment to evaluate whether foreign banks mitigate problems of related lending. We conjecture that firms created during the transition period are more likely to belong to cronies who established businesses in a moment of confusion to strip assets from the government. We find that when foreign bank presence becomes more pervasive these firms receive fewer loans and grow less. In contrast, foreign banks facilitate access to credit and foster growth of young companies born after the transition period. Perhaps more surprisingly, companies already existing before the transition period also receive more loans. This is most likely due to the fact that only the most viable businesses survived. Overall, these findings suggest that foreign bank entry helps mitigating problems of related lending.

Not only has foreign bank presence an impact on individual firm performance, but it also affects industrial structure. Foreign bank lending fosters entry and exit especially in bank dependent industries. This suggests that foreign banks are more willing to take hard choices than domestic banks, and confirms that foreign bank presence helps to mitigate connected lending problems. Even though foreign banks favor entry, lack of local knowledge remains a handicap. Indeed we find that small firms have a lower market share and a lower proportion of total assets in countries with stronger foreign bank presence.

A few studies have already analyzed the lending practices of foreign banks. Mian (2004) for example shows that foreign banks in Pakistan avoid lending to opaque firms, especially if the cultural and geographical distance between the CEO and the loan officer is large. Analogously, Berger, Klapper and Udell (2001) document that foreign banks in Argentina have difficulties lending to informationally opaque firms. Clarke, Cull and Soledad Martinez Peria (2001) and Clarke, Cull, Soledad Martinez Peria and Sanchez (2002), on the other hand, find that foreign banks lend to small firms at least as much as domestic banks do. Using survey data they further document that both small and large firms assess access to credit to ease following foreign bank entry. However, none of these papers has analyzed the actual impact of foreign bank integration on firm growth, capital structure, and investment policies. To the best of our knowledge our paper is the first to do so.

Our paper is related to a vast literature on finance and growth which following the lead of King and Levine (1993a, b) has analyzed how financial development in general and banking system development in particular affect growth in a large cross-section of countries. We evaluate different aspects of financial development, namely financial development induced by the integration of banking systems. Additionally, in contrast to most of the literature, we use firm level data (not macro data). In this respect, our paper is mostly related to recent studies that employ firm level data and analyze how different aspects of financial development affect firm growth and investment. In particular, Guiso, Sapienza and Zingales (2004) analyze the effect of financial development on firm growth, entry, and capital structure across Italian provinces. Similarly, Bertrand, Schoar and Thesmar (2004) analyze the effect of banking system deregulation on French firms and industrial structure. We complement their work by looking at the firm and industry level effects of a different aspect of a banking system, i.e., foreign bank presence.

We organize the rest of the paper as follows. Section II reviews the predictions regarding lending in emerging markets and foreign bank orientation, and presents recent empirical findings. Section III introduces the data and sample characteristics. Sections IV discusses the variables used in the specifications and displays and discusses the empirical results on firm growth and financing. Section V analyzes sectoral performance. Section VI concludes.

II. Theoretical Predictions on the Effects of Foreign Bank Entry in Eastern European Economies

In this Section we aim to highlight the possible benefits and drawbacks of foreign bank entry, in particular for Eastern European economies. In this way we strive to identify the channels through which foreign bank entry may affect firm growth and industrial structure, the main issue that we explore in the rest of the paper.

A. Credit Availability

Financial integration allows capital to flow from capital-abundant countries, where expected returns are low, to capital-scarce countries, where expected returns are high (Obstfeld and Rogoff (1995)). Capital inflows may foster growth by increasing the amount of funding available to domestic projects.

More in general, in countries with underdeveloped financial systems like the Eastern European economies, financial integration should increase the supply of finance and thus expand the national financial system of these countries. In this respect, financial integration is expected to spur faster growth across the board (Rajan and Zingales (1998), Guiso, et al. (2004)).

The beneficiaries of financial market integration may well depend on the nature of the capital flows. Wider availability of funds decreases the interest rate and the ensuing decrease in the cost of capital should abet all firms. Equity market liberalization on the other hand clearly benefit mainly listed companies or unlisted companies that are large enough to consider an IPO.

Since all firms borrow from banks, the benefits of foreign bank entry may well be more evenly distributed. Foreign bank presence fostering the development of the banking system widens the availability of credit and relaxes firm capital constraints also for small and young firms. Foreign bank presence may thus have pervasive positive effects on a country's level of entrepreneurial activity.

We expect that foreign bank entry might have been particularly beneficial for Eastern European economies. After the fall of the communist regimes, Eastern Europe badly needed capital to restructure its real economy. In particular, state-owned enterprises had to modernize to compete in competitive markets. Additionally, Eastern European economies badly needed new small firms to provide basic consumer goods and services, and entrepreneurs initially lacked access to start-up capital. But the Eastern European banking sector initially seemed inadequately small to satisfy this hefty demand for funds. For example, in 1993 domestic credit over GDP equaled around 55 percent in the transition countries in our sample and average bank assets per capita were below 1,300 US Dollars (Source: *IMF International Financial Statistics Yearbook*). In contrast, in the other 46 developing countries domestic credit over GDP actually exceeded 85 percent and average bank assets were above 1,500 US Dollars per capita. Bank assets in many developed European countries surpassed 40,000 US Dollars per capita. Foreign capital channeled by foreign banks contributed significantly to relax these constraints. By 1997 for example average bank assets in the transition countries had already increased to almost 2,000 US Dollars per capita.

B. Sounder Lending Practices

The ownership structure of domestic banks often leads to lending practices that are far from sound. Local governments and shareholders of non-financial companies often control domestic banks in developing countries. State or corporate control may give rise to conflicts of interests with pernicious effects on financial stability.

La Porta, et al. (2003) for example find that Mexican banks make larger loans at a lower interest rate to related companies that are then more likely to default. Similarly, state-owned banks are often driven by political considerations. Sapienza (2004) convincingly shows that in Italy loans from state-owned banks are a vehicle for supplying political patronage. Consistently, Mian (2003) finds that state-owned banks in emerging economies perform uniformly poorly and only survive due to strong government support.

Government ownership of banks is pervasive around the world, but particularly acute in Eastern European economies. La Porta, Lopez-de-Silanes and Shleifer (2002) for example estimate that governments control on average 40 percent of total bank assets, but in Eastern Europe governments still controlled almost 70 percent of all bank assets in the year 2000.

Problems of related lending seem also omnipresent in Eastern Europe. Laeven (2001) for example finds that banks in Russia often grant larger loans to companies that own equity in the bank. In addition, politicians in Eastern Europe continue to mobilize state-owned banks to support employment in state-owned or even recently privatized enterprises.²

Opening the domestic financial sector to foreign competition helps to mitigate these conflicts of interests. Domestic firms typically do not control foreign banks. While foreign governments own some foreign banks and these banks may be driven by political motives when lending to their respective home constituencies, these foreign state-owned banks are also naturally unencumbered by any domestic ownership ties and political motivations in making lending decisions.

For all these reasons, we expect foreign bank lending to stimulate firm growth and leveraging, not only because foreign banks may direct more capital into the country, but also because foreign bank presence may enhance allocational efficiency. Foreign banks are likely to shun businesses created during the transition years, because often these firms were mere conduits to strip assets from the government. There is actually evidence that domestic, in particular state-owned, banks favored transition businesses and in the process of privatization made large loans to potential entrepreneurs to enable them to tender and acquire firms (Simonson (2001)).

Small and young firms, a category particularly affected by the ineptitude and corruption of domestic bank officers (Beck, Demirguc-Kunt and Levine (2004b)), are expected to benefit most from foreign bank lending. Firms untainted by any past bank or state ownership ties are likely to be able to access to more bank loans and thus grow more if foreign bank presence increases. In addition, foreign bank lending should increase new firm creation and entry.

C. Hard versus Soft Information

Foreign banks may seek promising local projects and lend at fair rates rather than lending to related firms at below market loan rates. Foreign banks may also import lending expertise and sound lending practices. But foreign banks may suffer considerable organizational handicaps in engaging small and young local firms, a category of firms important for growth in developing countries.

Banks are often already sizeable before venturing abroad, following customers or seeking diversification (see the review by Clarke, Cull, Soledad Martinez Peria and Sanchez (2003)). Once abroad, they may cater to international companies from their home country, which seek their services (Berger, Dai, Ongena and Smith (2003)) and are often considered safer and more profitable borrowers. However, large banks may suffer from managerial diseconomies when engaging both relationship (small) and transactional (large) clients (Berger, Demsetz and Strahan (1999)).

Even more importantly, foreign banks may fail to collect "soft" information (for example, a character assessment of an entrepreneur, the degree of trust), which is crucial in lending to small firms. In fact, small and young firms typically report little or no "hard" information, for example detailed financial statements, credit history etc. (Berger and Udell (2002), Petersen (2002)). The use of soft information in lending decisions requires however a decentralized organization that grants local branch managers substantial decision powers (Liberti (2002)), because soft information cannot be passed as easily as hard information within the bank (Stein (2002)). Foreign banks may hesitate to decentralize because the local bank personnel may be considered lacking expertise or even untrustworthy.³

Some of these concerns may be mitigated by the fact that improvements in communication and information processing technology may have altered the possibilities to tap into, collect, and relay information on small businesses. Hence the range of firm opaqueness over which foreign banks are willing to fund may have expanded (Petersen and Rajan (2002)). Nevertheless, foreign bank presence may still hamper small and young firm financing and growth, in particular if foreign banks substitute for domestic banks, as we discuss in the next Section.

D. Competition, Stability, and Dynamic Effects in the Banking System

Even though access to credit for small and young firms may tighten when foreign bank presence is large, the net impact on these firms still need not be negative. Foreign bank presence may influence the banking system of a country in a number of different ways, such that small firms still end up benefiting. This is true even if no foreign banks would directly lend to small firms.

In developing countries, including Eastern European economies, foreign banks are often more efficient and profitable than domestic banks (Demirguc-Kunt and Huizinga (2000), Green (2003), Naaborg, Scholtens, de Haan, Bol and de Haas (2003)). Fostering competition, foreign banks may reduce profits and interest margins of all banks operating in the market (Claessens, Demirguc-Kunt and Huizinga (2001) and Unite and Sullivan (2003)).

In developing countries, foreign bank entry may also stabilize the financial system (Crystal, Dages and Goldberg (2002)). First, foreign banks have sounder lending practices and accumulate fewer bad loans. In addition foreign banks may be more resilient to negative shocks because of their direct access to foreign savings. On the other hand, foreign banks may introduce more volatility in lending because they can more easily find alternative investment opportunities (Morgan and Strahan (2003)) or transfer shocks from their home countries (Soledad Martinez Peria, Powell and Vladkova Hollar (2003)). However, the latter effect is likely to be second order in emerging markets that are generally exposed to significantly larger shocks than the foreign banks' home countries. Consistently, de Haas and Lelyveld (2003) find no evidence of increased instability following foreign bank entry for a set of transition countries. To the extent that foreign bank entry actually reduces concentration, fewer, not more, banking crises should ensue (Beck, Demirguc-Kunt and Levine (2004a)).

Finally, the mode of foreign bank entry may determine its effects on local financing. It is well known that if foreign banks enter through mergers and acquisitions, they have the potential to harm small local firms borrowing from the domestic target bank. Berger and Udell (1996) and Peek and Rosengren (1996) for example find that as domestic banks grow through consolidation, they tend to reduce the supply of loans to small businesses, in particular when the acquirer previously focused on large-firm lending (Peek and Rosengren (1998)).

On the other hand if foreign banks enter a new market by opening new branches they do not substitute domestic banks but simply increase the number of active financial intermediaries. Enhancing the development of the domestic banking system (as in II.A) without decreasing the number of financial intermediaries with local information can only be positive. This is also true if foreign banks enter by acquiring local distressed banks or state-owned banks, as has often been the case in

Eastern Europe. Distressed or state-owned banks were often plagued by ill-conceived and corrupted lending policies, and were unlikely to have played a major role in fostering local entrepreneurial activity in the first place.

Given the actual mode of entry of foreign banks in Eastern European economies it is not clear whether small firms were harmed considerably. In the first part of the nineties, foreign banks established primarily greenfield subsidiaries in Eastern Europe (de Haas and Lelyveld (2003)), increasing the level of financial intermediation without substituting domestic banks. When foreign banks acquired existing domestic banks they more often than not acquired banks in need of fresh capital, sometimes encouraged to do so by domestic regulation (to obtain a license in Poland for example, Naaborg, et al. (2003)). Foreign banks started only recently merging subsidiaries with domestic banks they already control, spurred by and contributing to an industry-wide global consolidation trend.

Whatever the mode of entry, even though the entrant or newly acquired foreign bank may focus on servicing predominantly large firms, incumbent or *de novo* domestic banks may step up the plate to fill the funding gap. Berger, Goldberg and White (2001) and Berger, Bonime, Goldberg and White (2004) show this to be the case in the US following domestic bank mergers that increased bank size and shifted the merged bank towards large business lending. Bonin and Abel (2000) provide anecdotal evidence that this dynamic effect may have moderated the impact of foreign bank entry in Hungary.

To conclude foreign bank entry may foster competition, efficiency, and stability, in which case firm growth and financing should increase across the board. On the other hand, small firm growth and financing may be negatively affected if foreign banks enter through M&As. In that case the net effect will also depend on the dynamic response by other competing banks.

III. Methodology and Identification

Identifying the effects of foreign bank entry is not an easy task and poses problems similar to the identification of the effects of financial development on growth. The mere correlation between financial development and growth cannot be interpreted as evidence of causality because financial markets may develop in the anticipation of future opportunities. Analogously, foreign banks may enter and lend to a larger extent in countries that are expected to grow more in the future.

We try to tackle this problem in different ways. First, we analyze the effect of foreign bank lending on firm rather than country growth. Looking at firm growth allows us to partially mitigate the problem of reverse causation because we are able to control for country fixed effects, time-varying growth opportunities, financial development, and GDP per capita.

Second, we can analyze the differential impact of foreign bank lending on firms with different characteristics (small and large firms, firms created before after and during the transition period). In this way, we test the validity of the channels through which foreign bank entry is expected to affect firm growth. Even if average firm growth and foreign bank lending were correlated because of an omitted common factor, it would be difficult to argue that such an omitted common factor affects the relation between foreign bank lending and firm growth in a systematic way for firms with different characteristics.

Third, and perhaps most convincingly, our results become stronger if we instrument foreign bank lending. During the sample period Eastern European countries pursued reforms that improved to varying degrees the protection of investor rights. We employ the creditor rights detailed in Pistor, Raiser and Gelfer (2000) as instruments. In particular, our instruments include: (1) creditors' control of the bankruptcy process, (2) creditors' control of the bankruptcy process, including reorganization consent, (3) the legal provisions on security interests, and (4) the *ex post* creditors' sanctions on management. Previous studies suggest that protection of creditor rights affects foreign bank lending. Esty (2003), for example, finds that different legal and financial systems affect the composition of loan syndicates. In

particular, foreign banks provide a greater share of total funds in countries with strong creditor rights, strong legal enforcement, and less-developed financial systems.

We use predetermined values of the institutional variables as is consistent with a causal link, and exploit changes in investor protection across countries to identify the effect of changes in foreign bank lending on our variable of interest.⁴ The intuition behind our identification strategy is similar to Jayaratne and Strahan (1996). They use the deregulation of bank branches in the U.S. as an instrument to show that improvements in the quality of bank lending are positively related to economic performance. Similarly, we analyze how the removal of implicit barriers to foreign bank presence –a weak institutional environment—affects economic performance.

To be able to interpret the relation between foreign bank presence and economic performance as a causal relation, we surmise that foreign banks did not influence the initial configuration of creditor rights or any later amendments.⁵ This is likely because foreign banks are not part of the domestic constituency the politicians want to please to be reelected. However, to establish the causal link, we also need that domestic banks and other economic agents did not influence creditor rights in a way that is systematically correlated to expected economic performance. In general, institutional change is never completely exogenous. The process of legal change in Eastern European economies however corroborates our assumptions. These countries started from very different initial conditions and exhibit a tendency to legal convergence Pistor (2000). Legal convergence seems to have been primarily the result of international institutions' technical assistance programs and of the harmonization requirements for countries wishing to join the European Union.

In addition, stronger creditor rights may both help and hurt domestic banks (as creditors and competitors to foreign banks respectively) and incumbents firms. The state of flux in the political process in Eastern Europe and the multitude of parties affected by changes in creditor rights complicated lobbying in a way that it makes arduous to posit and find a systematic link between economic performance and legal change, and particularly not given its timing and speed (which is the variation that we exploit to identify the effects of foreign bank lending). For these reasons, we believe that it is reasonable take legal change as exogenous.

We are aware that institutional characteristics may have a direct effect on growth for instance because they affect financial development. Desai, Gompers and Lerner (2003) for example show that country-specific political, legal, and regulatory variables influence entrepreneurial activity in Eastern European economies. However, Desai, et al. (2003) do not include creditor rights in their study and we further conjecture creditor rights may affect firm financing decisions foremost through its impact on foreign bank presence. Most importantly, we control for aggregate growth, GDP per capita, and in particular financial development, which are the alternative channels through which the institutional framework can affect firm growth.

Finally, we do not look at a single aspect of firm growth. We evaluate the impact of foreign lending on firm growth and look at the mechanisms through which foreign lending may affect growth. When observing a positive relationship between foreign lending and growth for a given category of firms, we can only interpret the correlation as causation if a mechanism consistent with such an interpretation – i.e., this category of firm increases the use of bank credit and decrease the use of alternative source of funds such as trade credit – is supported by the empirical evidence. Additionally, we also evaluate to what extent the results we find using firm level data are present in the aggregate sectoral data. All considered we are confident that our empirical methodology can provide evidence suggestive of a causal impact of foreign bank lending on the growth of entrepreneurial firms across different countries.

IV. Data and Sample Characteristics

A. Data Sources

We use data from a variety of sources. To construct our firm and sector specific variables we use the 2003 edition of *Amadeus* compiled by *Bureau Van Dijck*. Giannetti (2003) and recently Desai, et al. (2003) and Klapper, Laeven and Rajan (2004) also employ this dataset.

We extract firm-specific data for 14 Eastern European transition countries, listed in Table 1, for the years 1993 to 2002. The sample includes companies that meet at least one of the following three criteria: (1) its operating revenues are larger than or equal to ten million euros, (2) its book assets are larger than or equal to 20 million euros, and (3) the number of employees is larger than or equal to 150. The criteria are somewhat more restrictive for larger countries, in our sample the Russian Federation and Ukraine: cutoffs then equal 15 million, 30 million, and 200 respectively. Coverage of firm financial information expanded steadily throughout the sample period, but in particular from 1997 to 1998. For example, in 1993 we have information on the main balance sheet items for 1,673 firms, while in 2002 23,541 firms were covered.

To construct our bank sector variables we use the 2003 edition of *Bankscope*. We obtain GDP growth from the *World Development Indicators*, and, as explained in Section III, rely on Pistor, et al. (2000) for the creditor rights indices.

B. Descriptive Statistics

Table 1 reports sample characteristics by country. We report for each country the number of firms, foreign bank lending as a percentage of total bank lending, and average firm assets, age, and growth in assets in the year 2000 (a typical year for which coverage is optimal).

Our main proxy for foreign bank presence is the percentage of foreign lending (% Foreign Lending). We define % Foreign Lending as the ratio of loans extended by foreign banks to total bank loans in a given country. A bank is defined to be foreign if foreign individuals, corporations, financial institutions, or even foreign governments

combined own more than 50 percent of the bank. This cutoff is similar to the one used in previous literature (see, for instance, Mian (2003)) and reflects common majority voting rules. As the distribution of foreign ownership is highly bimodal, changing the cutoff will hardly affect the results. Indeed, 63 percent of all banks in the sample are 100 percent domestically owned. But foreigners own less than 50 percent in only 11 percent of the banks, while in almost 20 percent of the cases foreigners own more than 90 percent.

Foreign ownership is also more concentrated than domestic ownership. For example, the Herfindahl – Hirschman Index (HHI) (the sum of squared shares) of ownership concentration for domestic banks is only around 0.25, for foreign banks it is almost 0.75 (the difference is statistically significant at the 1 percent level). Hence, foreign banks are controlled by one or two foreign blockholders.

There is a large variation in foreign bank lending across the 14 countries and across time. The percentage foreign bank lending in 1996 for example ranges from 0 in the Republic of Macedonia to almost 92 percent in Bulgaria and across all countries foreign lending increases almost 10 percent in only four years, from 44 percent in 1996 to 53 percent in 2000.

In Table 1 we also categorize the countries by 1996 foreign lending into a high and low group (cutoff: 50 percent). Foreign lending in the low group increases faster. In addition, firm asset size and age are lower and asset growth is higher in the low group. The latter finding is particularly surprising in light of our earlier discussion but taken together with the empirical evidence on size and age demonstrates the value of investigating the differential impact of firm growth within each country.

We measure firm performance by sales and asset growth. As often argued, firm growth should be partly determined by the availability of credit. Some observations on firm sales seemed excessively large. To limit the influence of these outliers, we censored the growth rates at the 1 and 99 percentiles, admittedly *ad hoc* cutoffs. Given the many observations and controls in our empirical models, our key results should not be affected. Table 2 reports firm sales averaged across the sample in US Dollars.

We define sales growth as $ln(Sales_{t+1}/Sales_t)$ (and present it in percentage terms in all specifications). We denote this variable in the Tables as $\Delta ln(Sales)$. The

logarithm form should again contribute to minimizing the effects of the censored large values. Mean sales growth thus defined equals 11.3 percent. Similarly defined, mean asset growth equals 4.0 percent.

We further assess the effect of the availability of credit on the changes in the firms' capital structure by focusing on the increase in financial debt between t-I and t relative to the firm's total assets at time t (Δ Debt/Assets), and the increase in account payables between t-I and t relative to the firm's total sales at time t (Δ Trade Credit/Sales). Wider availability of credit should increase leverage, but decrease the use of trade credit. Consistently with this interpretation, the mean change in leverage equals 1.5 percent. The mean change in trade credit is -10.5 percent.

As indicated earlier foreign bank presence in a particular year in a country is measured as the percentage ratio of foreign bank to total bank lending. This variable is one of our main variables of interest. Foreign lending may enhance the availability and allocation of credit, increasing debt capacity (and therefore leverage) and stimulating growth. The mean percentage foreign lending equals 37.9 percent.

To analyze the differential effect of foreign bank presence on different categories of firms, we focus on three important firm characteristics: size, age, and efficiency. Firm *size* is a common measure of firm access to external funds and visibility. Smaller firms are typically expected to grow faster. However, to the extent that foreign banks have difficulties handling soft information or focus on large firm, small firm growth and ability to increase their debt may be stunted. We measure firm size by the logarithm of the number of employees. The mean (median) number of employees equals 645 (296).

Firm *age*, measured in years, commonly stands for the public track record of the firm, and is introduced in logarithmic form to capture the decreasing informational content of such a record as the firm ages. Younger firms are generally smaller, though the coefficient of partial correlation between the proxies for age and size in our sample is actually smaller than one percent. Like small firms, young firms are expected to grow faster. To the extent that foreign banks have difficulties handling private soft information, young firm growth and access to debt may be lower than for other firms. In addition, firm age in transition countries may proxy for the trustworthiness of the public track record. During the transition period that occurred in

those countries roughly between 1989 and 1993 many firms may have been started as a vehicle for asset stripping by dubious management. We call the firms that were created between 1989 and 1993 (the transition period) the "transition firms". Firms that started before 1989, on the other hand, though possibly trust worthier than the transition firms may have seen their public track record set to null and as a result may have been considered not unlike firms that started after 1993. To account for this non-monotonicity in age we also introduce dummies that equal one if the firm originated before or after the transition period respectively. The mean (median) age equals 18 (11) with 19 percent of the firms established before 1989 and 45 percent after 1993.

Finally, we introduce a measure of firm *efficiency*. Ex ante it is not entirely clear how efficiency will affect firm growth and financing. However, given their better lending practices and technology, foreign banks should have fewer problems finding and funding efficient firms. Moreover, foreign banks being less connected should favor efficient firms instead of related borrowers. Hence, foreign bank presence is expected to foster access to credit and growth for the most efficient firms. To construct a measure of firm efficiency we divide firm sales by the number of employees. We call a firm efficient when its sales per employee exceed that of the mean firm in its industry (first digit NACE), country, and year. According to this definition, 29 percent of the firms are classified as efficient.

In addition to the independent variables discussed above, we include a set of control variables. In all specifications we include up to 13 Country dummies to control for the fact that elements of a country's institutional and legal framework may affect firm growth and financing, as documented by Desai, et al. (2003) and Giannetti (2003). We also include up to 10 Industry and 9 Year dummies to control for industry and business cycle effects.

V. Results

A. Firm Growth

1. Benchmark Specifications

To assess the differential impact of foreign bank lending on firm growth, we start by regressing firm sales growth on foreign lending, firm characteristics, and country, industry, and year dummies. Next we instrument foreign lending using country specific measures of creditor rights and introduce the key interaction terms between foreign lending and firm characteristics.

We report the results in Table 3. We take a few natural steps to arrive at our empirical benchmark specification that is Model IV. In Model I we employ ordinary least squares, in Model II we instrument % Foreign Lending with the four creditor protection variables and add the efficiency and interaction dummies. In Model III we introduce the transition period dummies and in Model IV we add the ratio of total bank lending to GDP as a measure of financial development. We further correct all standard errors for clustering at the firm level. Taken together, the models illustrate the robustness of the estimated coefficients and the need to instrument our measure of foreign lending.

Our first-stage estimates confirm that the four legal protection variables have high explanatory power for foreign lending, as we can reject the null hypothesis that the four coefficients of legal protection variables equal zero at a 1 percent level of significance in a regression of foreign lending on the instruments and all other exogenous variables. In this respect, our instruments do not suffer from the problems of weak instruments described by Bound, Jaeger and Baker (1995).

The coefficients in all models suggest that foreign lending stimulates firm growth. The interaction terms we introduce in the various specifications in Table 4 suggest that small firms and more surprisingly more efficient and older firms benefit less from foreign bank entry. The fact that small firms benefit to a lesser extent from foreign banks suggests that inability to use soft information may indeed represent a handicap for foreign banks. It is at first sight more surprising that foreign banks do not

seem to convey loans to more efficient companies. The latter result however is not robust, economically quite small, and due with all probability to the definition of our proxy for efficiency. This variable, defined as sales per employee, most likely captures whether a firm business is close to the optimal size in terms of sales. This interpretation is consistent with the fact that efficient companies as well as older and large firms have lower growth rates.

The finding that older firms benefit less than younger companies is only apparently in contrast to the evidence that firms with lower degree of information asymmetry such as large firms receive fewer loans from foreign banks. This finding must be interpreted in the light of the experience of the Eastern European economies. Older firms in our sample are more likely to be born during the transition period and are to a large extent run by entrepreneurs who were able to enjoy the favors of politicians. The fact that they do not fully benefit from foreign bank entry simply suggests that foreign banks might be able to mitigate problems of related lending. This interpretation is confirmed by the fact that these companies appear to have worse corporate governance in our sample. Although it is difficult to define corporate governance in a sample that predominantly includes small unlisted companies, like ours, we have information on whether companies have attracted outside shareholders, an indication that they probably have a viable business and promise outside investors a reasonable return (Giannetti and Simonov (2004)). We find that companies born before and after the transition period have more dispersed ownership. In slightly more than 20 percent of them, the controlling shareholder controls less than 25 percent of the capital. In striking contrast, 44 percent of the companies born during the transition period have a shareholder controlling more than 25 percent of the capital. Most importantly, 45 percent of the companies born during the transition period have the state or a bank as a shareholder. Only 24 (21) percent of the companies born after (before) the transition period have the state or a bank as a shareholder. This indicates that problems of related lending may indeed be more pervasive for companies born during the transition period and that foreign banks help to cure these problems.

We further explore the conjecture that foreign banks discriminate against transition firms by including two dummies for firms born before and after the transition period (instead of firm age). We find that a higher percentage of foreign lending affects only the growth of firms born during the transition period –i.e. those firms with worse corporate governance – negatively and that firms born after 1993 but also the firms that were already in business before 1989 benefit from foreign bank presence. The pre-1989 firms that are still active are likely to be viable businesses. To this extent, these results suggest that foreign banks may enhance capital allocation.

2. Economic Relevancy

All, but one, reported coefficients in Model III are statistically significant at the 1 percent level. This significance is not surprising given the large number of observations (57,433) we employ. Hence assessing the economic relevance of the estimated coefficients is crucial. Table 4 reports such an assessment of the economic relevance of the various independent variables for sales growth. For easy reference we take the inverse logarithm of the calculated impacts.

Table 4 shows the impact on sales growth of an increase in foreign lending from 20 percent to 50 percent (approximately one half of a standard deviation on each side of the mean). This experiment would entail for example moving from Serbia and Montenegro to Hungary in 2002 or following the path of Romania from 1998 to 2002, of course all *ceteris paribus*. This 30 percent jump in foreign lending increases firm sales growth by almost 16 percent, a substantial effect (in the specification without interaction terms).⁷

The interaction terms suggest that however the effects of foreign bank lending are unevenly distributed across firms with different characteristics. As already indicated, foreign lending nurtures growth especially for large, non-transition, or inefficient firms. Firms larger than 300 employees (approximately the median) grow by more than 17 percent while firms smaller than this cutoff grow at a rate of only 15 percent.

Hence the picture that arises is that foreign bank lending in transition countries fosters firm growth, but that large firms benefit more. This effect is both statistically significant and economically relevant. We find these results in line with common fears ("small firms are hurt when foreign banks enter") but contrasting with the work by Clarke, et al. (2001) mentioned earlier. Their results indicate that the total effects were moderate, but they did not find significant differences between the impact on

small and large firm growth, possibly because of data and methodological issues. For instance, they were not able to fully control for differences in country growth opportunities as we do, and most importantly they evaluated the effects of foreign bank presence only through the entrepreneurs' declared ease in access to credit.

Foreign lending further fosters growth of the non-transition firms. If foreign bank lending increases for example from zero to 35 percent (the mean in our sample), pre-1989 and post-1993 firms grow by approximately 2 and 4 percent faster than the other firms, *ceteris paribus*. This finding suggests that related lending has a first-order effect on capital allocation and that foreign bank entry contributes significantly to mitigating this problem.

Finally, we also find that efficient firms grow slower and are adversely affected by foreign bank lending. However, in contrast to our other findings, this result is not robust, and may well depend on the fact that, as we note above, our measure of efficiency also captures optimality in scale.

3. Robustness

A possible critique to our interpretation of the results is that foreign bank presence and our instruments are correlated to some other factors we have not yet controlled for. For example the country and year fixed effects we include may not capture country time-varying growth opportunities. Foreign banks expanding their lending to be able to profit from the host country growth could explain the positive correlation between foreign bank presence and growth. Hence we control for the yearly country growth rate to capture a-synchronicities in business cycles. Although this variable is often positive and significant, our results remain qualitatively unchanged.

More problematic for our interpretation is that growth opportunities correlated with our proxy for foreign bank presence may differently affect the various categories of firms. We could for example observe that foreign banks expand their presence when growth opportunities improve while at the same time large firms grow faster, for reasons independent of their external financing arrangement. We already control for a wide-range of firm characteristics including industrial sector, firm size and age. However, it is plausible that some firms are able to expand sales and investment more

during a boom because of access to internal funds. For this reason we introduce a variable we call Firm Internal Growth that equals ROA / (1 - ROA) in all specifications. As usual, ROA is the firm's Return on Assets. Results again are virtually unaffected.

Next we explore if the identity of the lending banks matters. First, foreign banks could foster growth not because they are foreign, but simply because they are not state owned. To explore this possibility we include, the percentage of lending granted by private domestic banks or foreign banks. While this variable has generally a positive effect on growth, the effect of foreign bank lending remains positive and significant and, perhaps most importantly, larger from an economic point of view. Second, we explore whether foreign bank lending is correlated with financial development. Hence the positive effect of foreign bank lending on growth could merely reflect the fact that *more* credit is available, rather than *how* it is available. For this reason we include a measure of financial development, defined as total bank lending to GDP, in Model IV. Results are qualitatively almost unaffected. In unreported specifications we also interact our proxy for financial development with the firm characteristics that we employ to explore how the gains from foreign bank presence are distributed. Interestingly enough, financial development affects small and large firms equally, as the coefficient on the interaction term between firm size and financial development is generally not significant. However, this coefficient is mostly negative suggesting that, if anything, financial development ceteris paribus favors small firms, a result also found by Beck, Demirguc-Kunt, Laeven and Levine (2004).

We also find that financial development positively affects firm growth, as expected, but only if we do not control for the proportion of foreign lending. In interpreting this surprising finding we must keep in mind that our specification already includes country fixed effects. Hence we identify only the impact of changes in total loans to GDP on firm growth. Our results then indicate that an increase in total loans, in particular an increase of domestic loans to GDP, does not necessarily have a positive impact on firm growth. Our findings are consistent with empirical evidence showing that lending booms often result in an accumulation of bad loans (Gourinchas, Valdes and Landerretche (2001)) and that domestic banks frequently engage in

connected lending. To further explore this issue, we insert both the ratios of domestic loans to GDP and foreign loans to GDP in various specifications. The unreported estimates indicate that only increases in domestic loans to GDP are negatively related to firm growth in a consistent way.

Finally, we replace sales by asset growth and rerun all regressions. We report Model V in the last column of Table 3 and the corresponding economic relevancy tests in the third column in Table 4. All results are unaltered, except that now the coefficient on the interaction between foreign lending and the efficiency dummy becomes insignificant. The magnitudes of the impact on asset growth of changing the independent variables are surprisingly similar to the magnitude of the impact on sales growth.

B. Firm Financing

To investigate the mechanism underlying the results reported so far, we study the impact of foreign bank lending on firm financing. In particular, our interpretation of the empirical evidence on firm growth would be corroborated if we observed that firms – and in particular the firms that are observed to grow faster when foreign bank presence increases – make a larger use of financial loans if foreign banks expand lending. We first analyze growth in firm financial debt relative to total assets, defined as ((Debt_t - Debt_{t-1})/ Assets_t). Table 5 reports only one of the three steps we take to reach our benchmark specification, as the other two steps are not all that informative. We notice however that going from I to II that foreign lending no longer plays a direct role in affecting leverage growth and that all of the effect in Model II channels through the interaction terms.

As expected on the basis of our previous results, we find that foreign bank presence increases access to credit especially for large firms and non-transition firms. The economic effects remain sizeable. If foreign bank lending increases from 20 to 50 percent, small (large) firms increase their financial debt to asset growth by 0.6 (1) percent. Similarly, firms created before (after) the transition period increase financial debt to asset growth by 0.4 (0.8) percent.

Next, we run a similar set of robustness exercises as in the firm growth section, i.e. we consecutively add country growth, firm internal growth, and country

financial development. We report only the latter specification in Model III. Results are virtually unaffected, except that foreign lending again directly fosters leverage growth when financial development is added to the specification. Interesting, a higher level of credit to GDP does not imply higher leverage. Unreported estimates suggest that while an increase in foreign loans to GDP is related to higher leverage for all firms, this is not true for domestic loans to GDP, possibly suggesting that domestic banks train relatively more of their increase in lending towards the domestic public sector.

Firms appear not only to obtain access to more bank credit, but also the maturity of their liabilities increases (Model IV), especially for large and non-transition firms. Hence the widely held concern that foreign bank lending involves short-term "hot" money that is readably retracted during crises seems misplaced, at least for less risky and less opaque firms (Berger, Espinosa-Vega, Frame and Miller (2004), Ortiz-Molina and Penas (2004)). This finding is however also an indication that foreign bank presence may swing bank lending towards long-term transactional loans, as banks with strong relationships with borrowers generally offer short-maturity loans to be able to exercise control (Berger and Udell (1995)).

The increase in financial debt is also accompanied by a decrease in the cost of debt, defined as interest paid to total financial liabilities (Model V). Foreign banks appear to lower the interest rate in particular to firms without connections. Large firms, which were probably favored by state banks, experience a smaller decrease in the cost of debt.⁸

As financially constrained firms may make more use of trade credit (Petersen and Rajan (1994)), we expect that firms that benefit most from foreign bank entry in terms of growth and access to credit will also make less use of trade credit. To explore this conjecture, we analyze the changes in trade credit relative to sales ((Trade Credit_t - Trade Credit_{t-1})/ Sales_t) as a function of the same independent variables we employed so far in Model VI in Table 5. Indeed, we find that companies that are able to make greater use of bank loans when foreign bank presence increases also use less trade credit. This suggests that increased foreign bank presence contributes to relax financial constraints especially for the categories of firms we have identified. However the average effect of foreign lending on the variable measuring changes in

trade credit use is not significant in the specification (not reported) in which we do not include the interaction terms. This result should not come a surprise as all firms in our sample – even the ones that benefit most from foreign bank loans – are likely to be financially constrained (and consequently make large use of trade credit). From an economic point of view, the decrease in trade credit relative to sales due to an increase in foreign lending is however sizeable (Table 6 Model VI). A 20-to-50 increase in foreign lending reduces trade credit growth by 2 percent for transition firms, and by around 4 percent for non-transition firms.

To conclude, foreign lending increases access to foreign loans, relaxes credit constraints, and fosters firm growth. Foreign lending also improves allocational efficiency as the cost of debt decreases for firms without connection with banks (the non-transition firms) to a larger extent.

C. Sector Performance

In this Subsection, we assess the industry effects of foreign bank lending. This assessment is relevant for different reasons. First, this exercise allows us to evaluate the aggregate implications of foreign bank presence. In particular, we will be able to answer the question whether an increase in foreign lending affects firm entry, exit, and industrial structure. Answering this question allows us to further explore the channels through which foreign bank presence affects the economy. If increased foreign bank presence for example helps to mitigate problems of related lending, we expect that the exit rate is higher in countries with stronger foreign bank presence. Similarly, if foreign banks shun small firms, we expect that a country's industrial structure will be affected and that larger companies will command more market share and assets.

The sectoral analysis allows us to further scrutinize the validity of our identification strategy. We introduce a new instrument. From Barth, Caprio and Levine (2001), who compile an international database on commercial banking regulation, we glean the fraction of foreign banks' applications for commercial banking licenses that were rejected minus the fraction of domestic banks' rejected applications. This variable varies across countries but not across time. Hence it is too weak to function as an instrument in the firm level regressions, where we include both

time-series and cross-sectional variation (but control for country and year fixed effects). However, we can exploit this variable as an instrument in the industry level analysis, where, as is customary in the literature (e.g., Rajan and Zingales (1998)), we average across sector by country (but do not use the time-series variation). This new instrument, capturing the real and present barriers to entry for foreign banks, is even less likely to enter directly in the equations we estimate than the creditor protection variables. Employing a Hausman test we can thus use the new variable to test the validity of the creditor protection variables as instruments. In all cases we explored, we are not able to reject the null that the investor protection variables do not have an independent effect in the equation.

We investigate the impact of foreign lending on five sector characteristics: the number of firms, entrants, exits and the percentage of small firm sales and assets (defined as the share of sales and assets, respectively, of firms with employment below the median). We regress the logarithm of each one of these sector characteristics on the instrumented measure of foreign bank lending, and a measure of financial development. As in the previous subsection the latter measure is defined as the ratio of total bank lending to GDP in a country. We also include 74 industry dummies and control for the size of the sector in a given country with the level of employment at the beginning of the period. All values of the explanatory variables are taken at the beginning of the period while the dependent variable is a time average. Table 7 provides the coefficients, while Table 8 assesses their economic relevancy. The latter exercises are readily interpretable because the impacts are reported in level and at the means. The results further highlight the effects of foreign lending. Take entry and exit rates. Foreign bank lending seems to foster industry dynamics, as it stimulates both industry entry and exit. The effect is both statistically and economically significant as there are 15 (18) more entrants (exits) if the percentage of foreign lending increases from 20 to 50 percent. This is a large effect considering that the average number of entrant (exits) in a sector is 4.3 (2.6) and the standard deviation 12.3 (7.8). There is instead no statistically significant effect of foreign bank presence on the total number of firms. These results suggest that although foreign banks may avoid lending to small firms, which as a consequence invest and grow less, the problems possibly related to the foreign bank inefficiency in using local knowledge,

are not so severe as to discourage entry. Foreign lenders appear both more willing to engage entrants and to push exits.

The fact that financial development is not significantly correlated with exit (or in some unreported specifications even negatively correlated) – after controlling for the fraction of foreign loans – is not surprising in the light of the evidence showing that domestic banks are afflicted by related lending problems. This empirical evidence squares with our previous firm level results showing that foreign bank presence may help to cure these problems. It is more surprising instead that financial development is not significantly correlated with firm entry after controlling for foreign loans. Foreign banks thus appear to spur entrepreneurial activity, at least in countries where domestic banks lack lending expertise and do not have sound lending policies.

Finally, there is a dramatic difference in the way foreign lending and financial development may affect small firms' investment and market share in an industry. An increase in foreign lending from 20 to 50 percent of total loans decreases the proportion of sales (assets) of firms with employment below the median approximately by 37 (18) percent. Again the effect is sizeable as it explains more than one standard deviation of the variable. Hence, foreign lending substantially reduces the percentage of small firms' assets in an industry, while financial development has neither a statistically nor an economic significant effect, another vivid illustration of the important compositional effects of foreign bank lending.

Although the previous specifications allow us to quantitatively evaluate the economic impact of foreign lending, they are subject to the critique that the institutional variables we use to instrument foreign lending have a direct impact on growth. To further check whether the causal interpretation that we give of our estimates is warranted, we follow the methodology suggested by Rajan and Zingales (1998). Arguably, the effects of foreign bank presence should be larger in industries that depend more on bank loans. Similarly to Rajan and Zingales, we measure bank dependence in an industry with the ratio of financial loans to total liabilities. We can thus test whether the impact of foreign bank lending is larger in sectors that are more bank dependent by including an interaction variable between the proxy for bank dependence and foreign lending. Since our new variable of interest varies across sectors within a country, we are able to include country fixed effects that capture

unobserved country heterogeneity. The estimates show that our previous conclusions on the effects of foreign lending are confirmed: only in the equation for the small firms' proportion of sales our variable of interest is no longer statistically significant. From an economic point of view, our results are even more striking. Once we control for country fixed effects, it emerges that while foreign banks favor entry and exit in bank dependent sectors, financial development is negatively correlated with both.

The economic effect of foreign bank presence on entry and exit is halved when we include country fixed effects. This suggests that there is a country-specific component in industry turnover. Most importantly, foreign bank presence seems to be related to sectoral composition. Indeed, the number of firms in bank dependent sectors is significantly larger in countries with stronger foreign bank presence.

Overall, our sector analysis shows that foreign lending, industry churning, and large firm presence go hand in hand. Foreign lending improves credit allocation, but possibly to the detriment of small businesses' investment. Additionally, the quality of lending policies seems to matter significantly more for economic performance than the lending volumes. This finding is consistent with Jayaratne and Strahan (1996), who show that financial liberalization had positive effects in the U.S. not because an increase in the volume of credit but possibly because of improvements in bank efficiency.

VI. Conclusion

This paper analyzes how and to what extent the process of financial integration can benefit small entrepreneurial firms. In particular we focus on foreign bank lending. Banks represent an important source of capital for small firms. However small firms may not be able to benefit to the full extent from financial integration through foreign bank entry. Foreign banks may lack the local information that is particularly important for lending in countries where asymmetric information problems are severe and legal enforcement is weak. Additionally, foreign banks are often large organizations themselves and may be reluctant or unable to effectively use soft information. Soft information is often the only information available on small and young firms or potential entrepreneurs. Consequently, small firms may be able to benefit from financial integration to a lesser extent than larger and more established companies even if financial integration involves foreign bank entry.

Using a large data set of listed and unlisted companies in Eastern European economies, we find that foreign lending stimulates growth in firm sales, assets, and leverage, but that the effect is dampened for small firms. Even though foreign banks favor entry, lack of local knowledge remains a handicap. Indeed we find that small firms have a lower market share and a lower proportion of total assets in countries with stronger foreign bank presence.

Additionally, since we focus on Eastern European economies, we use the regime shift that took place between 1989-1993 as a natural experiment. We find that firms started during the transition period of 1989-1993, the ones which are more likely to have enjoyed politicians and connected banks' favors, benefit least from foreign bank entry. Foreign banks also increase exit especially in bank dependent industries. This confirms that foreign banks are more willing to take hard choices than domestic banks and thus mitigate connected lending problems.

While the impact and mechanism we identify in this paper seem robust and economically important, the effects we document probably provide a lower bound to the impact of foreign bank presence on industrial structure. In fact, the Eastern European economies became market economies only in the early nineties. Their

banking systems were largely underdeveloped and local banks lacked expertise in allocating loans. To this extent, the destruction of soft information due to acquisitions of domestic banks by foreign banks is likely to have been minimal. Arguably, the differential impact of foreign bank presence on large and small firms may be larger in countries where the acquired banks had a longer experience in extending credit to local firms.

Several other interesting questions remain unanswered. For example, do the mode of entry, the organizational form, the ownership structure, and the country of origin of the foreign banks operating in the country matter for the magnitude of the impact on the small firm sector? And does technological development and deeper economic and financial integration ultimately abate the effect of foreign bank presence on small firm growth and leverage? Finally, do foreign banks benefit firms only directly through their lending? Or do they also have positive effects on domestic banks' efficiency and foreign lending, which may indirectly benefit bank-dependent companies? We leave these questions for future research.

REFERENCES

- Acharya, V. V., R. K. Sundaram, and K. John, 2004, On the Capital Structure Implications of Bankruptcy Codes, (London Business School, London).
- Barth, J.R., G. Caprio, and R. Levine, 2001, The Regulations and Supervision of Banks Around the World: a New Database, (World Bank, Washington DC).
- Beck, T., A. Demirguc-Kunt, L. Laeven, and R. Levine, 2004, Finance, Firm Size, and Growth, (University of Minnesota).
- Beck, T., A. Demirguc-Kunt, and R. Levine, 2004a, Bank Concentration and Crises, (World Bank, Washington DC).
- Beck, T., A. Demirguc-Kunt, and R. Levine, 2004b, Financial and Legal Constraints to Firm Growth. Does Size Matter, *Journal of Finance* Forthcoming.
- Bekaert, G., C. Harvey, and C. Lundblad, 2003, Did Financial Liberalization Spur Economic Growth, (Columbia Business School, New York NY).
- Berger, A. N., S. D. Bonime, L. G. Goldberg, and L. J. White, 2004, The Dynamics of Market Entry: The Effects of Mergers and Acquisitions on Entry in the Banking Industry, *Journal of Business* Forthcoming.
- Berger, A. N., Q. Dai, S. Ongena, and D. C. Smith, 2003, To What Extent Will the Banking Industry be Globalized? A Study of Bank Nationality and Reach in 20 European Nations, *Journal of Banking and Finance* 27, 383-415.
- Berger, A. N., R. Demsetz, and P. Strahan, 1999, The Consolidation of the Financial Services Industry: Causes, Consequences, and Implications for the Future, *Journal of Banking and Finance* 23, 135-194.
- Berger, A. N., and R. DeYoung, 2001, The Effects of Geographic Expansion on Bank Efficiency, *Journal of Financial Services Research* 19, 163-184.
- Berger, A. N., M. Espinosa-Vega, S. Frame, and N. M. Miller, 2004, Debt Maturity, Risk, and Asymmetric Information, *Journal of Finance* Forthcoming.
- Berger, A. N., L. G. Goldberg, and L. J. White, 2001, The Effects of Dynamic Changes in Bank Competition on the Supply of Small Business Credit, *European Finance Review* 5, 115-139.
- Berger, A. N., I. Hasan, and L. F. Klapper, 2004, Further Evidence on the Link between Finance and Growth: An International Analysis of Community Banking and Economic Performance, *Journal of Financial Services Research* 25, 169-202.
- Berger, A. N., L. F. Klapper, and G. F. Udell, 2001, The Ability of Banks to Lend to Informationally Opaque Small Businesses, *Journal of Banking and Finance* 25, 2127-2167.
- Berger, A. N., N. M. Miller, M. A. Petersen, R. G. Rajan, and J. C. Stein, 2005, Does Function Follow Organizational Form? Evidence From the Lending Practices of Large and Small Banks, *Journal of Financial Economics* 76, 237-269.
- Berger, A. N., and G. F. Udell, 1995, Relationship Lending and Lines of Credit in Small Firm Finance, *Journal of Business* 68, 351-381.
- Berger, A. N., and G. F. Udell, 1996, Universal Banking and the Future of Small Business Lending, in A. Saunders, and W. Ingo, eds.: *Financial System Design: The Case for Universal Banking* (Irwin Publishing, Homewood IL).
- Berger, A. N., and G. F. Udell, 2002, Small Business Credit Availability and Relationship Lending: The Importance of Bank Organisational Structure, *Economic Journal* 112, 32-53.
- Bertrand, M., A. Schoar, and D. Thesmar, 2004, Banking Deregulation and Industry Structure: Evidence from the French Banking Reforms of 1985, (University of Chicago, Chicago IL).
- Bonin, J., and I. Abel, 2000, Retail Banking in Hungary: A Foreign Affair?, (Wesleyan University, Middletown CN).
- Booth, L. V., A. Aivazian, A. Demirguc-Kunt, and V. Maksimovic, 2001, Capital Structures in Developing Countries, *Journal of Finance* 56, 87-130.
- Bound, J., D. A. Jaeger, and R. M. Baker, 1995, Problems with Instrumental Variables: Estimation When the Correlation between the Instruments and the Endogenous Explanatory Variable is Weak, *Journal of the American Statistical Association* 90, 443-450.
- Chari, A., and P. B. Henry, 2004, Risk Sharing and Asset Prices: Evidence from a Natural Experiment, *Journal of Finance* 59, 1295-1324.

- Claessens, S., A. Demirguc-Kunt, and H. Huizinga, 2001, How Does Foreign Entry Affect Domestic Banking Markets, *Journal of Banking and Finance* 25, 891-911.
- Clarke, G., R. Cull, and M. Soledad Martinez Peria, 2001, Does Foreign Bank Penetration Reduce Access to Credit in Developing Countries? Evidence from Asking Borrowers, (World Bank, Washington DC).
- Clarke, G., R. Cull, M. Soledad Martinez Peria, and S. M. Sanchez, 2002, Bank Lending to Small Businesses in Latin America: Does Bank Origin Matter?, (World Bank, Washington DC).
- Clarke, G., R. Cull, M. Soledad Martinez Peria, and S. M. Sanchez, 2003, Foreign Bank Entry: Experience, Implications for Developing Economies, and Agenda for Further Research, *World Bank Research Observer* 18, 25-59.
- Crystal, J. S., B. G. Dages, and L. S. Goldberg, 2002, Has Foreign Bank Entry Led to Sounder Banks in Latin America?, *Current Issues in Economics and Finance* 8, 1-6.
- de Haas, R., and I. Lelyveld, 2003, Foreign Banks and Credit Stability in Central and Eastern Europe, (De Nederlandsche Bank, Amsterdam).
- Degryse, H., and S. Ongena, 2005, Distance, Lending Relationships, and Competition, *Journal of Finance* 60, 231-266.
- Demirguc-Kunt, A., and H. Huizinga, 2000, Determinants of Commercial Bank Interest Margins and Profitability: Some International Evidence, *World Bank Economic Review* 13, 379-408.
- Desai, M., P. Gompers, and J. Lerner, 2003, Institutions, Capital Constraints and Entrepreneurial Firm Dynamics: Evidence from Europe, (National Bureau of Economic Research, Cambridge MA).
- Esty, B. C., 2003, When Do Foreign Banks Finance Domestic Investment? New Evidence on the Importance of Legal and Financial Systems, (Harvard Business School, Boston MA).
- Giannetti, M., 2003, Do Better Institutions Mitigate Agency Problems? Evidence from Corporate Finance Choices, *Journal of Financial and Quantitative Analysis* 38, 185-212.
- Giannetti, M., and A. Simonov, 2004, Which Investors Fear Expropriation? Evidence from Investors' Portfolio Choices, (Stockholm School of Economics, Stockholm).
- Gourinchas, P. O., R. O. Valdes, and O. Landerretche, 2001, Lending Booms: Latin America and the World, *Economia* 1, 47-99.
- Green, C. J., 2003, Are Foreign Banks in Central and Eastern Europe More Efficient Than Domestic Banks, (Loughborough University, Loughborough).
- Guiso, L., P. Sapienza, and L. Zingales, 2004, Does Local Financial Development Matter?, *Quarterly Journal of Economics* Forthcoming.
- Henry, P. B., 2000a, Do Stock Market Liberalizations Cause Investment Booms, *Journal of Financial Economics* 58, 301-334.
- Henry, P. B., 2000b, Stock Market Liberalization, Economic Reform, and Emerging Market Equity Prices, *Journal of Finance* 55, 529-564.
- Henry, P. B., 2003, Capital Account Liberalization, The Cost of Capital, and Economic Growth, *American Economic Review* 93, 91-96.
- Jayaratne, J., and P. E. Strahan, 1996, The Finance-Growth Nexus: Evidence from Bank Branch Deregulation, *Quarterly Journal of Economics* 111, 639-670.
- King, R. G., and R. Levine, 1993a, Finance and Growth: Schumpeter May be Right, *Quarterly Journal of Economics* 108, 713-737.
- King, R. G., and R. Levine, 1993b, Finance, Entrepreneurship and Growth, *Journal of Monetary Economics* 32, 513-542.
- Klapper, L. F., L. Laeven, and R. G. Rajan, 2004, Business Environment and Firm Entry: Evidence from International Data, (World Bank, Washington DC).
- Kroszner, R. S., and P. E. Strahan, 1999, What Drives Deregulation? Economics and Politics of the Relaxation of Bank Branching Restrictions, *Quarterly Journal of Economics* 124, 1437-1467.
- La Porta, R., F. Lopez-de-Silanes, and A. Shleifer, 2002, Government Ownership of Banks, *Journal of Finance* 57, 265-301.
- La Porta, R., F. Lopez-de-Silanes, A. Shleifer, and R. W. Vishny, 1998, Law and Finance, *Journal of Political Economy* 106, 1113-1155.
- La Porta, R., F. Lopez-de-Silanes, and G. Zamarripa, 2003, Related Lending, *Quarterly Journal of Economics* 128, 231-268.
- Laeven, L., 2001, Insider Lending and Bank Ownership: The Case of Russia, *Journal of Comparative Economics* 29, 207-229.

- Levine, R., 2004, Finance and Growth: Theory and Evidence, (National Bureau of Economic Research, Cambridge MA).
- Liberti, J. M., 2002, Initiative, Incentives and Soft Information: How Does Delegation Impact the Role of Bank Relationship Managers?, (University of Chicago, Chicago IL).
- Mian, A., 2003, Foreign, Private Domestic, and Government Banks: New Evidence from Emerging Markets, (University of Chicago, Chicago).
- Mian, A., 2004, Distance Constraints: The Limits of Foreign Lending in Poor Economies, (University of Chicago, Chicago).
- Morgan, D. P., and P. E. Strahan, 2003, Foreign Bank Entry and Business Volatility: Evidence from US States and Other Countries, (National Bureau of Economic Research, Cambridge MA).
- Naaborg, I., B. Scholtens, J. de Haan, H. Bol, and R. de Haas, 2003, How Important Are Foreign Banks in the Financial Development of European Transition Countries, (CESifo, Munchen).
- Obstfeld, M., and K. Rogoff, 1995. *Foundation of International Macroeconomics* (MIT University Press, Cambridge MA).
- Ortiz-Molina, H., and M. F. Penas, 2004, Lending to Small Businesses: The Role of Loan Maturity in Addressing Information Problems, (University of British Columbia, Vancouver BC).
- Peek, J., and E. S. Rosengren, 1996, Small Business Credit Availability: How Important Is the Size of Lender?, in A. Saunders, and W. Ingo, eds.: *Financial System Design: The Case for Universal Banking* (Irwin Publishing, Homewood IL).
- Peek, J., and E. S. Rosengren, 1998, Bank Consolidation and Small Business Lending: It's Not Just Bank Size That Matters, *Journal of Banking and Finance* 22, 799-820.
- Petersen, M. A., 2002, Information: Hard and Soft, (Northwestern University, Chicago IL).
- Petersen, M. A., and R. G. Rajan, 1994, The Benefits of Lending Relationships: Evidence from Small Business Data, *Journal of Finance* 49, 3-37.
- Petersen, M. A., and R. G. Rajan, 2002, Does Distance Still Matter? The Information Revolution in Small Business Lending, *Journal of Finance* 57, 2533-2570.
- Pistor, K., 2000, Patterns of Legal Change: Shareholder and Creditor Rights in Transition Economies, European Business Organization Law Review 2, 59-79.
- Pistor, K., M. Raiser, and S. Gelfer, 2000, Law and Finance in Transition Economies, (European Bank for Reconstruction and Development, London).
- Rajan, R.G., and L. Zingales, 1998, Financial Dependence and Growth, *American Economic Review* 559-586.
- Sapienza, P., 2004, The Effects of Government Ownership on Bank Lending, *Journal of Financial Economics* 72, 357-384.
- Simonson, D. G., 2001, Foreign Bank Influence in the Czech Republic, in R.E. Litan, P. Masson, and M. Pomerleano, eds.: *Open Doors. Foreign Participation in Financial Systems in Developing Countries* (Brookings Institution Press, Washington D.C.).
- Soledad Martinez Peria, M., A. Powell, and I. Vladkova Hollar, 2003, Banking on Foreigners: The Behavior of International Bank Lending to Latin America, 1985-2000, (World Bank, Washington DC).
- Stein, J., 2002, Information Production and Capital Allocation: Decentralized versus Hierarchical Firms, *Journal of Finance* 57, 1891-1922.
- Strahan, P., 2004, Comment on Berger, Hasan, and Klapper, *Journal of Financial Services Research* 25, 203-206.
- Unite, A. A., and M. J. Sullivan, 2003, The Effect of Foreign Entry and Ownership Structure on the Philippine Domestic Banking Market, *Journal of Banking and Finance* 27, 2323-2345.

TABLE 1. SAMPLE CHARACTERISTICS

The table reports by country the Number of Firms, Foreign Bank Lending, Firm Assets, Age, and Asset Growth in the indicated year.

By Country	2000 Number	1996 Foreign	2000 Foreign	2000	2000	2000 Firm
by Country	of Firms	Bank Lending	Bank Lending	Firm Assets	Firm Age	Asset Growth
		In %	% uI	In Million US\$	In years	% uI
BULGARIA	1,661	91.6	63.2	9.1	30.0	10.7
CROATIA	1,134	61.4	70.7	34.0	32.5	9.6
CZECH REPUBLIC	3,995	53.3	76.0	22.7	11.1	12.1
ESTONIA	501	49.0	6.86	15.1	15.1	16.2
HUNGARY	2,354	45.4	58.7	24.2	10.4	22.4
LATVIA	594	40.7	41.5	13.0	6.6	11.3
LITHUANIA	354	81.3	92.1	21.5	10.0	13.3
POLAND	7,487	61.0	65.0	28.9	13.6	9.2
REPUBLIC OF MACEDONIA	335	0.0	0.0	91.2	31.9	1.2
ROMANIA	3,318	4.1	52.5	13.2	7.6	22.1
RUSSIAN FEDERATION	22,042	17.1	10.8	13.7	7.8	8.9
SLOVAK REPUBLIC	1,134	82.2	94.7	85.1	11.4	0.5
SLOVENIA	805	25.8	21.7	40.3	17.2	3.5
UKRAINE	7,239	3.7	4.4	13.6	19.7	11.6
Average		44.0	53.5	30.4	16.5	10.9
Average 1996 Foreign Lending > 50%	2,627	71.8	77.0	33.6	18.1	9.2
Average 1996 Foreign Lending < 50%	4,648	23.2	36.1	28.0	15.2	12.2

TABLE 2. VARIABLE DEFINITIONS

The table reports the Variable Names, Definitions, data *Source* (*S*), Unit (U), Mean, Standard Deviation (SD), Minimum (Min), and Maximum (Max) for the main variables. The sample includes the maximum number of Observations (Obs) available or used in the estimations from 1993 to 2002. Data Sources (S) include: *Amadeus* (A), *Bankscope* (B), *Barth*, et al. (2001) (C), *Pistor*, et al. (2000) (P), and *World Development Indicators* (W). The Units (U) used are: percentage (%), thousands (T), years (Y), and millions of US Dollars (\$).

Variable Groups Variable Names	Definition	S	U	Mean	SD	25%	%0\$	75%	Obs
Dependent Firm									
Sales	Firm sales	A	∽	184.1	6304	_	3	6	57,453
Δln(Sales)	$= \ln(\text{Sales}_{l'} \setminus \text{Sales}_{l-1})$	A	%	11.3	57.4	-111	8	30	57,453
Assets	Firm assets	A	⇔	231.6	6151.8	_	3	11	63,593
Aln(Assets)	$= \ln(Assets_l/Assets_{t-1})$	A	%	4.0	44.7	-15	_	20	63,593
Debt / Assets	Ratio of firm financial debt to total assets	A	%	175.9	0.8689	0	5	20	45,994
ΔDebt / Assets	$= (Debt_t - Debt_{t-1}) / Assets_t$	A	%	2.6	22.1	<u>6</u> -	0	\mathcal{S}	45,994
Trade Credit / Sales	Ratio of firm payables to total assets	A	%	191.2	52.9	9	13	27	44,475
ΔTrade Credit / Sales	= (Trade Credit, - Trade Credit,) / Sales,	A	%	-10.5	139.8	-5	0-	4	44,475
LT Debt / Debt	Ratio of firm long-term financial liabilities to total financial liabilities	A	%	46.2	49.0	0	40	86	30,233
ALT Debt / Debt	$= (LT Debt_t - LT Debt_{t-1}) / Debt_t$	A	%	1.5	37.2	-10	0	6	30,233
Interest / Debt	Ratio of firm interest payments to total financial liabilities	A	%	23.3	17.3	8	17	32	34,827
In(Interest / Debt)	$= \ln(\operatorname{Interest}_t / \operatorname{Debt}_t)$	A	%	19.7	15.9	7	16	28	34,827
Dependent Sector									
# Firms	Number of firms in sector i, country j, year t	A		0.99	215.6	4	13	46	827
# Entrants	Ratio of number entrants to number of firms in sector i , country j , year t	A		4.3	12.6	0	1	4	827
# Exits	Ratio of number of exits to number of firms in sector i , country j , year t	A		2.6	7.8	0	1	7	827
% Small Firm Assets	Ratio of assets of firms below median in number of employees to total assets in sector <i>i</i> , country <i>j</i> , year <i>t</i>	A	%	6.89	21.0	56	73	85	827
% Small Firm Sales	Ratio of sales of firms below median in number of employees to total sales in sector <i>i</i> , country <i>j</i> , year <i>t</i>	A	%	51.7	27.8	33	59	72	827

Indopendent Country
Ratio of foreign loans to total bank lending in country j, pear t B % 37.9 29.2 4 45 57
Ratio of foreign loans to total bank lending in country j , g
Ratio of foreign loans to total bank lending in country j , g
Ratio of foreign loans to total bank lending in country j , g
Ratio of foreign loans to total bank lending in country j , g
Ratio of foreign loans to total bank lending in country j , g
Ratio of foreign loans to total bank lending in country j, year t Ratio of foreign loans to total bank lending in country j, year t Ratio of foral bank lending to GDP in country j, year t B % 17.7 15.9 6 10 31
Ratio of foreign loans to total bank lending in country j, year t Ratio of total bank lending in country j, year t Ratio of total bank lending to GDP in country j, year t Ratio of total bank lending to GDP in country j, year t Ratio of total bank lending to GDP in country j, year t Ratio of formancial loans to total liabilities in sector i A W W W W W W W W W
Ratio of foreign loans to total bank lending in country j, gear t Ratio of foreign loans to total bank lending in country j, year t Ratio of total bank lending to GDP in country j, year t Ratio of foral bank lending to GDP in country j, year t Ratio of foral bank lending to GDP in country j, year t Ratio of financial loans to total liabilities in sector i A W W W W W W W W W
Ratio of foreign loans to total bank lending in country j , gear t grant Ratio of foreign loans to total bank lending in country j , year t GDP growth in country j , year t W % 3.0 14.1 -3 4 16 GDP growth in country j , year t W % 3.1 10 25 0 30 Number of employees in sector i A T 0.8 11.7 0 0 1 Firm number of employees in sector i A - 645.3 1848 200 296 557 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1
Ratio of foreign loans to total bank lending in country j, page to sear to total bank lending in country j, year to total bank lending to GDP in country j, year to total bank lending to GDP in country j, year to total bank lending to GDP in country j, year to total labilities in sector i
Ratio of foreign loans to total bank lending in country j , g
ent Ratio of foreign loans to total bank lending in country <i>j</i> , gear <i>t</i> Ratio of total bank lending to GDP in country <i>j</i> , year <i>t</i> GDP growth in country <i>j</i> , year <i>t</i> Ratio of financial loans to total liabilities in sector <i>i</i> Number of employees in sector <i>i</i> Number of employees in sector <i>i</i> A T 0.8 117 0.5 30 34 Firm number of employees in sector <i>i</i> A T 0.8 1.7 0.8 1.7 0.0 0 1 Firm number of employees in sector <i>i</i> A dummy variable that equals one if the firm number of A - 0.51 0.50 0.7 2.2 2 A dummy variable that equals one if the firm started A - 0.19 0.40 0 0 0 A dummy variable that equals one if the firm started after A - 0.45 0.50 0.50 0 0 1
Ratio of foreign loans to total bank lending in country j , g
The section of foreign loans to total bank lending in country j , B , ϕ
ent Ratio of foreign loans to total bank lending in country j , B % 37.9 29.2 4 45 57 year t and Ratio of total bank lending to GDP in country j , year t M % 3.0 14.1 -3 4 16 GDP growth in country j , year t M % 3.1 10 25 30 34 Number of employees in sector i M % 3.1 10 25 30 34 16 Firm number of employees, i
The stip of foreign loans to total bank lending in country j , k
Ratio of foreign loans to total bank lending in country j , B % 37.9 29.2 4 45 57 year t and Ratio of total bank lending to GDP in country j , year t W % 3.0 14.1 -3 4 16 GDP growth in country j , year t W % 3.0 14.1 -3 4 16 Number of employees in sector i A
Ratio of foreign loans to total bank lending in country j , B % 37.9 29.2 4 45 57 year t and Ratio of total bank lending to GDP in country j , year t B % 17.7 15.9 6 10 31 GDP growth in country j , year t W % 3.0 14.1 -3 4 16 Number of financial loans to total liabilities in sector t t % 31 10 25 30 34 Number of employees in sector t
The stiff of foreign loans to total bank lending in country j , g
The stiff of foreign loans to total bank lending in country j , g
Ratio of foreign loans to total bank lending in country j , B % 37.9 29.2 4 45 57 year t Ratio of foreign loans to total bank lending in country j , year t GDP growth in country j , year t W % 3.0 17.7 15.9 6 10 31 GDP growth in country j , year t W % 3.0 14.1 -3 4 16 Number of financial loans to total liabilities in sector i A 7 0.8 11.7 0.8 25 30 34 Number of employees in sector i Firm number of employees, A - 645.3 1848 200 296 557 A dummy variable that equals one if the firm number of A - 0.51 0.50 0 1 Firm age A 7 17.8 20.8 9 11 13
Ratio of foreign loans to total bank lending in country j , B % 37.9 29.2 4 45 57 year t and Ratio of total bank lending to GDP in country j , year t M % 3.0 17.7 15.9 6 10 31 GDP growth in country j , year t M % 3.0 14.1 -3 4 16 Number of financial loans to total liabilities in sector i A % 31 10 25 30 34 Number of employees in sector i A 7 0.8 1.7 0.8 1.7 0 0 1 1 Firm number of employees, i
Ratio of foreign loans to total bank lending in country j , B % 37.9 29.2 4 45 57 year t ent Ratio of total bank lending to GDP in country j , year t GDP growth in country j , year t Katio of financial loans to total liabilities in sector i Number of employees in sector i Firm number of employees, A
Ratio of foreign loans to total bank lending in country j , B % 37.9 29.2 4 45 57 year t and the firm number of employees; is larger than 300 (modian).
Ratio of foreign loans to total bank lending in country j , B % 37.9 29.2 4 45 57 year t and Ratio of total bank lending to GDP in country j , year t 8 % 17.7 15.9 6 10 31 GDP growth in country j , year t 7 % 3.0 14.1 -3 4 16 Number of financial loans to total liabilities in sector i 8 % 31 10 25 30 34 Number of employees in sector i 8 % 7 0.8 11.7 0 0 1 1 Firm number of employees i 8 % 557 i 8 i 9 % 1.3 i 9 % 6 6 A dummy variable that equals one if the firm number of i 9 % 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1
Ratio of foreign loans to total bank lending in country j , B % 37.9 29.2 4 45 57 year t Ent Ratio of total bank lending to GDP in country j , year t GDP growth in country j , year t Katio of financial loans to total liabilities in sector i Number of employees in sector i Firm number of employees, $A = \frac{1}{2}$ $A = \frac{1}{2}$ $A = \frac{1}{2}$ $A = \frac{1}{2}$ Firm number of employees, $A = \frac{1}{2}$ A
Ratio of foreign loans to total bank lending in country j , B % 37.9 29.2 4 45 57 year t and Ratio of total bank lending to GDP in country j , year t B % 17.7 15.9 6 10 31 GDP growth in country j , year t W % 3.0 14.1 -3 4 16 Number of financial loans to total liabilities in sector i A % 31 10 25 30 34 Number of employees in sector i A T 0.8 1.7 0 0 1 1 Firm number of employees i
Ratio of foreign loans to total bank lending in country j , B % 37.9 29.2 4 45 57 year t and Ratio of total bank lending to GDP in country j , year t B % 17.7 15.9 6 10 31 GDP growth in country j , year t W % 3.0 14.1 -3 4 16 Number of employees in sector i A 7 0 0.8 1.7 0 0 1 Firm number of employees A 7 - 645.3 1848 200 296 557
Ratio of foreign loans to total bank lending in country j , B % 37.9 29.2 4 45 57 year t Ratio of total bank lending to GDP in country j , year t B % 17.7 15.9 6 10 31 GDP growth in country j , year t W % 3.0 14.1 -3 4 16 Ratio of financial loans to total liabilities in sector i A
Ratio of foreign loans to total bank lending in country j , B % 37.9 29.2 4 45 57 year t Ratio of total bank lending to GDP in country j , year t B % 17.7 15.9 6 10 31 GDP growth in country j , year t W % 3.0 14.1 -3 4 16 Ratio of financial loans to total liabilities in sector i A % 31 10 25 30 34 Number of employees in sector i A T 0.8 1.7 0 0 0 1
Ratio of foreign loans to total bank lending in country j , B % 37.9 29.2 4 45 57 year t Ratio of total bank lending to GDP in country j , year t B % 17.7 15.9 6 10 31 GDP growth in country j , year t W % 3.0 14.1 -3 4 16 Ratio of financial loans to total liabilities in sector i A % 31 10 25 30 34 Number of employees in sector i A T 0.8 1.7 0 0 1
Ratio of foreign loans to total bank lending in country j , B % 37.9 29.2 4 45 57 year t Ratio of total bank lending to GDP in country j , year t B % 17.7 15.9 6 10 31 GDP growth in country j , year t W % 3.0 14.1 -3 4 16 Ratio of financial loans to total liabilities in sector i A % 31 10 25 30 34 Number of employees in sector i A T 0.8 1.7 0 0 1
Ratio of foreign loans to total bank lending in country j , B % 37.9 29.2 4 45 57 year t Ratio of total bank lending to GDP in country j , year t B % 17.7 15.9 6 10 31 GDP growth in country j , year t W % 3.0 14.1 -3 4 16 Ratio of financial loans to total liabilities in sector i A
Ratio of foreign loans to total bank lending in country j , B % 37.9 29.2 4 45 57 year t Ratio of total bank lending to GDP in country j , year t B % 17.7 15.9 6 10 31 GDP growth in country j , year t W % 3.0 14.1 -3 4 16 Ratio of financial loans to total liabilities in sector i A % 31 10 25 30 34
Ratio of foreign loans to total bank lending in country j , B % 37.9 29.2 4 45 57 year t Ratio of total bank lending to GDP in country j , year t B % 17.7 15.9 6 10 31 GDP growth in country j , year t W % 3.0 14.1 -3 4 16 Ratio of financial loans to total liabilities in sector i A % 31 10 25 30 34
Ratio of foreign loans to total bank lending in country j , B % 37.9 29.2 4 45 57 year t Ratio of total bank lending to GDP in country j , year t B % 17.7 15.9 6 10 31 GDP growth in country j , year t W % 3.0 14.1 -3 4 16
Ratio of foreign loans to total bank lending in country j , B % 37.9 29.2 4 45 57 year t Ratio of total bank lending to GDP in country j , year t B % 17.7 15.9 6 10 31 GDP growth in country j , year t W % 3.0 14.1 -3 4 16
Ratio of foreign loans to total bank lending in country j , B % 37.9 29.2 4 45 57 year t Ratio of total bank lending to GDP in country j , year t B % 17.7 15.9 6 10 31 GDP growth in country j , year t W % 3.0 14.1 -3 4 16
Ratio of foreign loans to total bank lending in country j , B % 37.9 29.2 4 45 57 year t Ratio of total bank lending to GDP in country j , year t B % 17.7 15.9 6 10 31 GDP growth in country j , year t W % 3.0 14.1 -3 4 16
Ratio of foreign loans to total bank lending in country j , B % 37.9 29.2 4 45 57 year t Ratio of total bank lending to GDP in country j , year t W % 3.0 17.7 15.9 6 10 31 GDP growth in country j , year t W % 3.0 14.1 -3 4 16
Ratio of foreign loans to total bank lending in country j , B % 37.9 29.2 4 45 57 year t Ratio of total bank lending to GDP in country j , year t B % 17.7 15.9 6 10 31 GDP growth in country j , year t W % 3.0 14.1 -3 4 16
Ratio of foreign loans to total bank lending in country j , B % 37.9 29.2 4 45 57 year t Ratio of total bank lending to GDP in country j , year t 8 % 17.7 15.9 6 10 31 GDP growth in country t year t W % 3.0 14.1 -3 4 16
Ratio of foreign loans to total bank lending in country j , B % 37.9 29.2 4 45 57 year t Ratio of total bank lending to GDP in country j , year t B % 17.7 15.9 6 10 31
Ratio of foreign loans to total bank lending in country j , B % 37.9 29.2 4 45 57 year t Ratio of total bank lending to GDP in country j , year t 8 % 17.7 15.9 6 10 31
Ratio of foreign loans to total bank lending in country j , B % 37.9 29.2 4 45 57 year t
Ratio of foreign loans to total bank lending in country j , B % 37.9 29.2 4 45 57 year t
Ratio of foreign loans to total bank lending in country j , B % 37.9 29.2 4 4.5 57
Ratio of foreign loans to total bank lending in country j , p 0, 27 0 20.0 A 1, A 5 57
1-1-1 -1 -1 -1 -1 -1 -1 -1 -1 -1 -1 -1 -
Jonandent Country

Instrumental									
	Creditor's control of the bankruptcy process	Ь		3.7	1.1	4	4	4	63,593
	Creditor's control of the bankruptcy process, including p reorganization consent	\boldsymbol{b}	1	3.3	1.3	8	4	4	63,593
	Legal provisions on security interests	Ь	1	1.3	8.0	1	-	7	63,593
	Ex post creditors' sanctions on management	\boldsymbol{b}	,	1.1	9.0	0.7	1	7	63,593
	Proportion of rejected foreign bank licenses minus proportion of rejected domestic banks licenses		% 2	-14	22	-17	0	0	774

TABLE 3. FIRM GROWTH

The table reports the coefficients and significance levels from ordinary least squares (Model I) and instrumental variable (Models II to V) estimations. Standard errors that are corrected for clustering at the firm level are reported in parentheses. The dependent variables are the % growth rate in the log of firm Assets in Model V. The definition of the variables can be found in Table 2. All specifications include up to 14 Country, 10 Industry, and 9 Year Dummies. *, **, and *** = significant at 10%, 5% and 1% level, two-tailed.

Model	П	II	III	VI	Λ
Dependent Variable	Δln(Sales)	Δln(Sales)	Δln(Sales)	Δln(Sales)	Δln(Assets)
Number of Observations	57,453	57,453	57,453	57,453	63,593
% Foreign Lending	0.87	*** 99.0	0.38 ***	1.17 ***	0.55 ***
)	(0.02)	(0.03)	(0.03)	(0.05)	(0.02)
ln(Firm Employees)	-9.46	*** 59.6-	-9.41 ***	-8.92 ***	-9.23 ***
	(0.28)	(0.37)	(0.37)	(0.37)	(0.28)
ln(Firm Age)	*** 57.7-	-5.25 ***	***	-5.76 ***	-6.71 ***
	(0.44)	(0.45)	(0.54)	(0.52)	(0.32)
d(Efficient Firm)		-5.20 ***	-5.24 ***	-7.65 ***	-0.21
		(1.41)	(1.39)	(1.28)	(0.87)
% Foreign Lending * d(Firm Employees)		0.10 ***	0.07	*** \ \ \ 0.0	0.10 ***
		(0.01)	(0.01)	(0.01)	(0.01)
% Foreign Lending * d(Firm Age)		-0.19 ***			
		(0.01)			
% Foreign Lending * d(Efficient Firm)		*** 80.0-	** \(\text{-0.0-} \)	-0.02	0.02
		(0.03)	(0.03)	(0.02)	(0.02)
% Foreign Lending * d(Firm Before 1989)			0.25 ***	0.11 ***	0.16 ***
			(0.02)	(0.01)	(0.01)
% Foreign Lending * d(Firm After 1993)			0.22 ***	0.24 ***	0.11 ***
			(0.01)	(0.01)	(0.01)
Financial Development				-0.40 ***	
				(0.02)	
Constant	102.12 ***	61.02 ***	69.64 ***	60.42 ***	71.68 ***
	(12.94)	(13.08)	(13.62)	(13.71)	(8.08)
R squared	0.11	0.12	0.12	90.0	0.19

TABLE 4. IMPACT OF FOREIGN LENDING ON FIRM GROWTH

table. All other variables are set equal to their means. For easy reference we take the inverse log of the calculated impacts and we also repeat the significance levels on the respective coefficients. The dependent variables are the % growth rate in firm Sales and Assets. The definition of the variables can be found in Table 2. For easy reference The table reports the percentage change in the dependent variable as a result of the indicated change in independent variables in Models III and V reported in the previous we also repeat the significance levels on the respective coefficients. *, **, and *** = significant at 10%, 5% and 1% level, two-tailed.

Model	III	Λ
Dependent Variable	ΔSales _t / Sales _{t-1}	$\Delta Assets_{t'} Assets_{t-1}$
If % Foreign Lending increases from 20 to 50%		
% Foreign Lending	13.6 ***	19.0 ***
% Foreign Lending * d(Firm Employees)	1.2 ***	1.6 ***
% Foreign Lending * d(Efficient Firm)	** 5.0-	0.3
% Foreign Lending * d(Firm Before 1989)	1.5 ***	1.0 ***
% Foreign Lending * d(Firm After 1993)	3.4 ***	1.6 ***

TABLE 5. FIRM FINANCING

reported in parentheses. The dependent variables are the % growth rate in firm Debt/Assets in Models I to III, the % growth rate in firm Trade Čredit/Sales in Model IV, the % growth rate in firm Long-Term Debt/Debt in Model V, and the % log of firm Interest payment/Debt in Model VI. The definition of the variables can be found in Table 2. All snecifications include in to 14 Country. 10 Industry, and 9 Year Dummies. The table reports the coefficients and significance levels from instrumental variable estimations. Standard errors that are corrected for clustering at the firm level are

Model	I	II	III	Λ I	Λ	IV
Dependent Variable	ΔDebt/Assets	ΔDebt/Assets	ΔDebt/Assets	ALTDebt/Debt	In(Interest/Debt)	ΔTrade/Sales
Number of Observations	45,994	45,994	45,994	30,233	34,827	44,475
% Foreign Lending	0.10 ***	0.01	*** 80.0	0.13 ***	-0.04 ***	-0.07
0	(0.01)	(0.01)	(0.01)	(0.02)	(0.01)	(0.05)
ln(Firm Employees)	-2.16 ***	-2.14 ***	-2.09 ***	-1.72 ***	0.19	4.96 ***
	(0.15)	(0.15)	(0.15)	(0.25)	(0.12)	(0.77)
In(Firm Age)	-0.83 ***	-1.52 ***	-1.41 ***	-2.51 ***	0.45 **	3.88 ***
	(0.13)	(0.14)	(0.14)	(0.36)	(0.18)	(0.85)
d(Efficient Firm)	1.89 ***	1.93 ***	1.60 ***	4.92 ***	-2.63 ***	-0.90
	(0.46)	(0.46)	(0.45)	(96.0)	(0.39)	(2.22)
% Foreign Lending * d(Firm Employees)	0.02 ***	0.02 ***	0.02 ***	0.02 **	0.01 **	-0.16 ***
	(0.00)	(0.00)	(0.00)	(0.01)	(0.00)	(0.04)
% Foreign Lending * d(Firm Age)	-0.06 *** (0.00)					
% Foreign Lending * d(Efficient Firm)	-0.02 ***	-0.02	-0.01	*** 80.0-	0.03 ***	0.19 **
	(0.01)	(0.01)	(0.00)	(0.02)	(0.00)	(0.07)
% Foreign Lending * d(Firm Before 1989)		*** 90.0	0.04 ***	*** 90.0	*** 80.0-	-0.38 ***
		(0.00)	(0.00)	(0.01)	(0.00)	(0.04)
% Foreign Lending * d(Firm After 1993)		0.05 ***	0.05 ***	0.03 ***	***	-0.14 ***
		(0.00)	(0.00)	(0.01)	(0.00)	(0.02)
Financial Development			-0.02 ***			
•			(0.00)			
Constant	7.65 ***	14.59 ***	16.53 ***	26.28 ***	7.58	-55.37 ***
	(2.49)	(3.02)	(3.00)	(5.76)	(5.78)	(13.31)
Regulared	0.07	0.07	0.07	0.03	60 0	0.21

TABLE 6. IMPACT OF FOREIGN LENDING AND FINANCIAL DEVELOPMENT ON FIRM FINANCING

previous table. All other variables are set equal to their means. For easy reference we take the inverse log of the calculated impacts and we also repeat the significance levels on the respective coefficients. The dependent variables are the % growth rate in firm Debt/Assets, Trade Credit/Sales, and Long-Term Debt/Debt and the change in The table reports the percentage change in the dependent variable as a result of the indicated change in independent variables in Models II, IV, V, and VI reported in the the % firm Interest payment/Debt respectively. The definition of the variables can be found in Table 2. For easy reference we also repeat the significance levels on the respective coefficients. *, **, and *** = significant at 10%, 5% and 1% level, two-tailed.

Model	П	2	>	VI
Dependent Variable	ΔDebt/Assets	ALTDebt/Debt	Δ (Interest/Debt)	ΔTrade/Sales
I.				
If % Foreign Lending increases from 20 to 50%				
% Foreign Lending	9.0	4.1 ***	*** 5.1-	-2.1
% Foreign Lending * d(Firm Employees)	** **		0.3 **	-2.3 ***
% Foreign Lending * d(Efficient Firm)	-0.1			
% Foreign Lending * d(Firm Before 1989)	0.4 **	0.4 ***	*** 9.0-	-2.0 ***
% Foreign Lending * d(Firm After 1993)				

TABLE 7. SECTOR PERFORMANCE

Entrants, and Exits and the log of the percentage Sales or Assets by Small Firms. The definition of the variables can be found in Table 2. The lower panel reports the change in the number of Firms, Entrants, and Exits and the percentage Sales or Assets by Small Firms at the means of these variables as a result of the indicated change in the independent variables. *, **, and *** = significant at 10%, 5% and 1% level, two-tailed. industry dummies, the specifications in the lower panel in addition include up to 15 country dummies. The dependent variables are the log of number of Firms, Both panels report the coefficients and significance levels from instrumental variable (IV) estimations. The specifications in the upper panel include up to 74

			1111	<u>\</u>	>
Dependent Variable	# Firms	# Entrants	# Exits	% Small Firm Sales	% Small Firm Assets
% Foreign Lending	0.01	0.04 ***	*** LO:0	-4.12 **	-1.02 ***
)	(0.03)	(0.01)	(0.01)	(1.73)	(0.21)
Financial Development	0.00	-0.00	0.00	0.02	0.02
•	(0.01)	(0.00)	(0.00)	(0.12)	(0.04)
Sector Employment	0.12 **	0.01	-0.01 *	-5.75	-1.75
	(0.06)	(0.04)	(0.04)	(5.32)	(10.23)
Constant	-0.20	-2.70 **	-6.24 ***	-168.72 **	-33.38 ***
	(0.60)	(1.04)	(1.24)	(78.67)	(10.71)
Rsquared	0.25	0.24	0.47	0.04	0.19
Number of Observations	269	269	269	642	267
% Foreign Lending * Financial Dependence	0.10 ***	*** 60.0	0.10 ***	-4.97	*** 69.8-
	(0.03)	(0.03)	(0.03)	(4.31)	(1.53)
Financial Development * Financial Dependence	-0.01 ***	-0.01 ***	-0.01 ***	1.28 *	6.17 ***
•	(0.00)	(0.00)	(0.00)	(0.71)	(0.00)
Sector Employment	-0.01 *	-0.01 ***	-0.10 ***	-0.90	0.78
	(0.00)	(0.00)	(0.03)	(4.91)	(1.60)
Constant	-2.25 ***	** 66.0	08.0	-377.55 **	-45.86 ***
	(0.43)	(0.52)	(0.59)	(128.88)	(22.10)
R squared	0.81	0.87	82.0	0.39	0.36
N	017	649	610	557	7

TABLE 8. IMPACT OF FOREIGN LENDING AND FINANCIAL DEVELOPMENT ON SECTOR PERFORMANCE

Both panels report the change in the dependent variable as a result of the indicated change in independent variables in Models I to V reported in the previous table. All other variables are set equal to their means. For easy reference we take the inverse log of the calculated impacts and we also repeat the significance levels on the respective coefficients. The dependent variables are the number of Firms, Entrants, and Exits and the percentage Sales or Assets by Small Firms. The definition of the variables can be found in Table 2. For easy reference we also repeat the significance levels on the respective coefficients. *, **, and *** = significant at 10%, 5% and 1% level, two-tailed.

Model	I	II	III	VI	VI
Dependent Variable	# Firms	# Entrants	# Exits	% Small Firm Sales	% Small Firm Assets
If % Foreign Lending increases from 20 to 50%					
% Foreign Lending	52.6	15.0 ***	18.0 ***	-37.1 **	-17.8 ***
If Financial Development increases from 10 to 25%					
Financial Development	12.5	7.0-	0.3	2.2	2.7
If % Foreign Lending increases from 20 to 50%					
% Foreign Lending * Financial Dependence	124.0 ***	7.8 **	4.9 ***	-25.1 **	-37.6 ***
If Financial Development increases from 10 to 25%					
Financial Development * Financial Dependence	-24.9 ***	***	-1.0 ***	8.6	4.0 ***

NOTES

¹ Levine (2004) provides a comprehensive review of the literature. ² See Simonson (2001) for evidence on the Czech Republic.

³ Berger and DeYoung (2001) study the effect of physical distance on bank branch control. Lending to small firms across large distances and borders is less common (Berger, Miller, Petersen, Rajan and Stein (2005)) and possibly less profitable for the bank (Degryse and Ongena (2005)).

⁴ This is a direct consequence of the inclusion of country fixed effects.

⁵ Kroszner and Strahan (1999) argue that U.S. state level deregulation of restrictions on bank branching and interstate banking, the instrument used by Jayaratne and Strahan (1996), was influenced by small-bank financial health and hence success in lobbying. See also Strahan

⁶ Our definitions minimize the impact of changes in assets on leverage and of changes in sales on trade credit availability.

Mean sales growth is 11 percent. In Berger, Hasan and Klapper (2004) an increase in Foreign Share, defined as the market share held by foreign-owned banks, from 20 to 50 percent raises GDP growth by between 1 and 2.5 percent. Mean GDP growth between 1994 and 2000 for the 28 developing countries in their sample equals 3 percent.

⁸ Interestingly, if we consider the ratio of financial expenses (instead of only the interest paid) and total financial liabilities, we observe that these increase when foreign bank lending increases. This suggests that foreign banks may offer more expensive financial services to firms

European Central Bank working paper series

For a complete list of Working Papers published by the ECB, please visit the ECB's website (http://www.ecb.int)

- 448 "Price-setting behaviour in Belgium: what can be learned from an ad hoc survey?" by L. Aucremanne and M. Druant, March 2005.
- 449 "Consumer price behaviour in Italy: evidence from micro CPI data" by G. Veronese, S. Fabiani, A. Gattulli and R. Sabbatini, March 2005.
- 450 "Using mean reversion as a measure of persistence" by D. Dias and C. R. Marques, March 2005.
- 451 "Breaks in the mean of inflation: how they happen and what to do with them" by S. Corvoisier and B. Mojon, March 2005.
- 452 "Stocks, bonds, money markets and exchange rates: measuring international financial transmission" by M. Ehrmann, M. Fratzscher and R. Rigobon, March 2005.
- 453 "Does product market competition reduce inflation? Evidence from EU countries and sectors" by M. Przybyla and M. Roma, March 2005.
- 454 "European women: why do(n't) they work?" by V. Genre, R. G. Salvador and A. Lamo, March 2005.
- 455 "Central bank transparency and private information in a dynamic macroeconomic model" by J. G. Pearlman, March 2005.
- 456 "The French block of the ESCB multi-country model" by F. Boissay and J.-P. Villetelle, March 2005.
- 457 "Transparency, disclosure and the Federal Reserve" by M. Ehrmann and M. Fratzscher, March 2005.
- 458 "Money demand and macroeconomic stability revisited" by A. Schabert and C. Stoltenberg, March 2005.
- 459 "Capital flows and the US 'New Economy': consumption smoothing and risk exposure" by M. Miller, O. Castrén and L. Zhang, March 2005.
- 460 "Part-time work in EU countries: labour market mobility, entry and exit" by H. Buddelmeyer, G. Mourre and M. Ward, March 2005.
- 461 "Do decreasing hazard functions for price changes make any sense?" by L. J. Álvarez, P. Burriel and I. Hernando, March 2005.
- 462 "Time-dependent versus state-dependent pricing: a panel data approach to the determinants of Belgian consumer price changes" by L. Aucremanne and E. Dhyne, March 2005.
- 463 "Break in the mean and persistence of inflation: a sectoral analysis of French CPI" by L. Bilke, March 2005.
- 464 "The price-setting behavior of Austrian firms: some survey evidence" by C. Kwapil, J. Baumgartner and J. Scharler, March 2005.
- 465 "Determinants and consequences of the unification of dual-class shares" by A. Pajuste, March 2005.
- 466 "Regulated and services' prices and inflation persistence" by P. Lünnemann and T. Y. Mathä, April 2005.
- 467 "Socio-economic development and fiscal policy: lessons from the cohesion countries for the new member states" by A. N. Mehrotra and T. A. Peltonen, April 2005.

- 468 "Endogeneities of optimum currency areas: what brings countries sharing a single currency closer together?" by P. De Grauwe and F. P. Mongelli, April 2005.
- 469 "Money and prices in models of bounded rationality in high inflation economies" by A. Marcet and J. P. Nicolini, April 2005.
- 470 "Structural filters for monetary analysis: the inflationary movements of money in the euro area" by A. Bruggeman, G. Camba-Méndez, B. Fischer and J. Sousa, April 2005.
- 471 "Real wages and local unemployment in the euro area" by A. Sanz de Galdeano and J. Turunen, April 2005.
- 472 "Yield curve prediction for the strategic investor" by C. Bernadell, J. Coche and K. Nyholm, April 2005.
- 473 "Fiscal consolidations in the Central and Eastern European countries" by A. Afonso, C. Nickel and P. Rother, April 2005.
- 474 "Calvo pricing and imperfect common knowledge: a forward looking model of rational inflation inertia" by K. P. Nimark, April 2005.
- 475 "Monetary policy analysis with potentially misspecified models" by M. Del Negro and F. Schorfheide, April 2005.
- 476 "Monetary policy with judgment: forecast targeting" by L. E. O. Svensson, April 2005.
- 477 "Parameter misspecification and robust monetary policy rules" by C. E. Walsh, April 2005.
- 478 "The conquest of U.S. inflation: learning and robustness to model uncertainty" by T. Cogley and T. J. Sargent, April 2005.
- 479 "The performance and robustness of interest-rate rules in models of the euro area" by R. Adalid, G. Coenen, P. McAdam and S. Siviero, April 2005.
- 480 "Insurance policies for monetary policy in the euro area" by K. Küster and V. Wieland, April 2005.
- 481 "Output and inflation responses to credit shocks: are there threshold effects in the euro area?" by A. Calza and J. Sousa, April 2005.
- 482 "Forecasting macroeconomic variables for the new member states of the European Union" by A. Banerjee, M. Marcellino and I. Masten, May 2005.
- 483 "Money supply and the implementation of interest rate targets" by A. Schabert, May 2005.
- 484 "Fiscal federalism and public inputs provision: vertical externalities matter" by D. Martínez-López, May 2005.
- 485 "Corporate investment and cash flow sensitivity: what drives the relationship?" by P. Mizen and P. Vermeulen, May 2005.

- 486 "What drives productivity growth in the new EU member states? The case of Poland" by M. Kolasa, May 2005.
- 487 "Computing second-order-accurate solutions for rational expectation models using linear solution methods" by G. Lombardo and A. Sutherland, May 2005.
- 488 "Communication and decision-making by central bank committees: different strategies, same effectiveness?" by M. Ehrmann and M. Fratzscher, May 2005.
- 489 "Persistence and nominal inertia in a generalized Taylor economy: how longer contracts dominate shorter contracts" by H. Dixon and E. Kara, May 2005.
- 490 "Unions, wage setting and monetary policy uncertainty" by H. P. Grüner, B. Hayo and C. Hefeker, June 2005.
- 491 "On the fit and forecasting performance of New-Keynesian models" by M. Del Negro, F. Schorfheide, F. Smets and R. Wouters, June 2005.
- 492 "Experimental evidence on the persistence of output and inflation" by K. Adam, June 2005.
- 493 "Optimal research in financial markets with heterogeneous private information: a rational expectations model" by K. Tinn, June 2005.
- 494 "Cross-country efficiency of secondary education provision: a semi-parametric analysis with non-discretionary inputs" by A. Afonso and M. St. Aubyn, June 2005.
- 495 "Measuring inflation persistence: a structural time series approach" by M. Dossche and G. Everaert, June 2005.
- 496 "Estimates of the open economy New Keynesian Phillips curve for euro area countries" by F. Rumler, June 2005.
- 497 "Early-warning tools to forecast general government deficit in the euro area: the role of intra-annual fiscal indicators" by J. J. Pérez, June 2005.
- 498 "Financial integration and entrepreneurial activity: evidence from foreign bank entry in emerging markets" by M. Giannetti and S. Ongena, June 2005.

